

INSTITUTO POTOSINO DE INVESTIGACIÓN CIENTÍFICA Y TECNOLÓGICA, A.C.

POSGRADO EN CONTROL Y SISTEMAS DINÁMICOS

Coexistence of stable states in a biparametric family of bimodal maps.

Tesis que presenta

Francisco Alejandro Jiménez Valdivia

Para obtener el grado de

Maestro en Control y Sistemas Dinámicos

Director de Tesis:

Dr. Eric Campos Cantón

San Luis Potosí, S.L.P., julio de 2024



Constancia de aprobación de la tesis.

La tesis **Coexistence of stable states in a biparametric family of bimodal maps** presentada para obtener el Grado de Maestro en Control y Sistemas Dinámicos en la opción control y sistemas dinámicos fue elaborada por **Francisco Alejandro Jiménez Valdivia** y aprobada el **dd de mm de 2024** por los suscritos, designados por el Colegio de Profesores de la División de Control y Sistemas Dinámicos del Instituto Potosino de Investigación Científica y Tecnológica, A.C..

> Dr. Eric Campos Cantón (Director de la tesis)

Dr. Cesar Octavio Maldonado Ahumada (Sinodal de la tesis)

Dr. Haret Codratian Rosu (Sinodal de la tesis)

Dr. Juan Gonzalo Barajas Ramírez (Sinodal de la tesis)



Créditos Institucionales.

Esta tesis fue elaborada en la División de Control y Sistemas Dinámicos A.C. del Instituto Potosino de Investigación Científica y Tecnológica, A.C., bajo la dirección del Dr. Eric Campos Cantón.

Durante la realización del trabajo el autor recibió una beca académica del Consejo Nacional de Humanidades, Ciencias y Tecnologías (No. de registro 1241841) y del Instituto Potosino de Investigación Científica y Tecnológica, A. C. Copia del acta del examen

A mis padres, Ma. Alejandra Valdivia y Francisco Jiménez, quienes con su incondicional apoyo, formación y valores, han sido la base fundamental para que esto sea posible. A mi querido asesor, Dr. Eric Campos Cantón, cuya orientación y guía han sido indispensables en este camino. Sin ellos, este logro no habría sido alcanzado.

Agradecimientos.

- Quiero expresar mi enorme gratitud por el valioso conocimiento, apoyo y mentoría proporcionados por el Dr. Eric Campos Cantón en el desarrollo de este trabajo. Sin su invaluable contribución, este proyecto no habría sido posible.
- Agradezco enormemente al Dr. Cesar Maldonado, cuyo estímulo y orientación despertaron mi interés en el área de los sistemas dinámicos discretos. También le agradezco sinceramente por aceptar ser mi sinodal, su apoyo ha sido fundamental en este proceso.
- Quiero agradecer al Dr. Gonzalo Barajas Ramírez por su invaluable apoyo, por compartir su vasto conocimiento y por aceptar amablemente ser mi sinodal. Su presencia siempre llena de carisma ha sido un verdadero motor de inspiración en cada paso de este proyecto.
- Agradezco al Dr. Haret Codratian Rosu por aceptar ser mi sinodal y por compartir su valioso conocimiento en este trabajo.
- Agradezco infinitamente a mi asesor de licenciatura, el Dr. Kayim Pineda Urbina, por su total apoyo en mis decisiones. Su confianza en mis capacidades y su constante aliento para perseguir mis sueños han sido invaluables.
- Agradezco profundamente a mis padres, a mi amada madre María del Carmen Castillo y a mi querido padre Mateo Valdivia, quienes siempre me brindaron su incondicional apoyo y bendición. Aunque partieron antes de este logro, sé en lo más profundo de mi corazón que, estén donde estén, ellos estarán orgullosos de mí. Los amo para siempre.

- Agradezco el apoyo de cada uno de mis queridos compañeros de maestría: Fernando Flores, Juan Manuel Cárdenas, Javier Quí y Alberto de La Rosa. Sin su ayuda en todos los aspectos, este logro tampoco habría sido posible.
- Agradezco profundamente a mi querida compañera, Paulina Elizabeth Vera Torres, por su constante apoyo en cada paso de esta maestría. Gracias a su dedicación y acompañamiento, este logro fue posible.
- Quiero expresar mi profundo agradecimiento a mi mejor amigo, Carlos Fernando Díaz Torres, quien siempre me alentó en cada decisión que tomé. Siempre creyó en mí, incluso más de lo que yo podía creer en mí mismo.
- Finalmente, pero no menos importante, quiero expresar mi más sincero agradecimiento a la Señora Marta Elena Cuello Garza. Su enorme y maravilloso apoyo fue fundamental en este logro. Sin su ayuda, definitivamente esto no habría sido posible.

Contents

Resumen				
Al	Abstract			
1	Intr	oduction	1	
	1.1	Multimodals maps	1	
	1.2	Monostability, bistability, trapping region and invariant set in dynamical systems	5	
	1.3	Motivation	9	
	1.4	Objective	10	
	1.5	Organization	10	
2	Mat	thematical preliminaries	12	
	2.1	Topology of the Real Numbers	12	
	2.2	Linear Algebra	16	
	2.3	Mathematical Analysis	17	
	2.4	Discrete-time Dynamical Systems	19	
3	Bist	ability in a bimodal map	25	
4	Fan	nilies of bimodal maps	53	

4.1	The monoparametric family of maps for the parameter β_2	53
4.2	The monoparametric family of maps for the parameter β_1 $~$.	65
4.3	The monoparametric family of maps for the parameter $\gamma~$	72

5 Conclusion

86

List of Figures

1.1	The logistic map for different values of the parameter α	3
1.2	The Tent map for different values of the parameter μ	3
1.3	Family of multimodal maps based on the logistic map [1]	4
1.4	Difference map showing different values of the parameter β .	4
1.5	Image first presented in 1915 with the title ' My Wife and My Mother-in-law', where the bistability in visual perception is demonstrated [2]	6
1.6	Bistability in the energetic description of a system	7
1.7	Logistic map for different values of $\alpha \in (0, 4] \subset \mathbb{R}$: (a) $\alpha = 3.8$ and (b) $\alpha = 4$	8
4.1	Bimodal maps of the monoparametric family f_D (4.1) for dif- ferent values of β_2 . The black line corresponds to $\beta_2 = 2$, the red line $\beta_2 = 3$, and finally, the blue one $\beta_2 = 4$	54
4.2	Cobweb diagram for the bimodal map f_D with $\beta_2 = 3$ using the initial condition $x_0 = 0.68 \in \mathcal{A}_2(3)$	58
4.3	Orbit of the bimodal map f_D with $\beta_2 = 3$ using the initial condition $x_0 = 0.68 \in \mathcal{A}_2(3)$, where it's possible to observe the convergence of the trajectory towards the attractor fixed point $x^* = \frac{3}{2}$	50
	point $x_{2,1} - \frac{1}{4}$	09
4.4	Cobweb diagram for the bimodal map f_D with $\beta_2 = 3$ with initial condition $x_0 = 0.85$ such that $x_0 \notin \mathcal{A}_2(3)$	59

different initial conditions $x_0 \in \mathcal{A}_2(3.3)$; $x_0 = 0.87$	60
4.6 Orbit of the bimodal map f_D with $\beta_2 = 3.3$ using the initial condition $x_0 \in \mathcal{A}_2(3.3)$; $x_0 = 0.87$, where it's possible to observe a period-two orbit.	60
4.7 Cobweb diagram for the bimodal map f_D with $\beta_2 = \frac{3 + \sqrt{17}}{2}$	
using the initial condition $x_0 = 0.61 \in \mathcal{A}_2\left(\frac{3+\sqrt{17}}{2}\right)$	62
4.8 Cobweb diagram for the bimodal map f_D with $\beta_2 = 3.57$ using the initial condition $x_0 = 0.61 \in \mathcal{A}_2(3.57)$.	63
4.9 Bifurcation diagram of the bimodal map f_D given by (4.1) with respect to the parameters β_1 and β_2 , where $\beta_1 = 2$ and $\beta_2 \in (0, 4] \subset \mathbb{R}$.	64
4.10 Bimodal map f_D (4.4) with $\gamma = 4$, $\beta_2 = 3.56$ and different values of β_1 . The black line represents $\beta_1 = 2$, the red line $\beta_1 = 3$, and finally, the blue one $\beta_1 = 4$	66
4.11 (a) Cobweb diagram and (b) orbit of the bimodal map f_D with $\beta_1 = 2, \ \beta_2 = 3.56$ and $x_0 = 0.0357 \in \mathcal{A}_1$.	69
4.12 (a) Cobweb diagram and (b) orbit of the bimodal map f_D with $\beta_1 = 2, \ \beta_2 = 3.56$ and $x_0 = 0.6557 \in \mathcal{A}_2(3.56)$.	70
4.13 (a) Cobweb diagram and (b) orbit of the the bimodal map f_D with $\beta_1 = 2.5$ and $\beta_2 = 3.56$ using the initial condition $x_0 = 0.9502 \notin \mathcal{A}_2(3.56)$	70
4.14 Bifurcation diagram of the bimodal map f_D given by (4.4) with respect to the parameters β_1 and β_2 , where $\beta_2 = 3.56$ and $\beta_1 \in (0, 4]$.	71
4.15 Bimodal map f_D (4.5) with $\gamma = 4$, $\beta_1 = \beta_2 = 4$ and different values of $\gamma \in (0, 4]$. The black line represents $\gamma = 2$, the red line $\gamma = 3$, and finally, the blue one $\gamma = 4$.	72
4.16 Cobweb diagram for the bimodal map f_D for: a) $\gamma = 2$ using the initial condition $x_0 = 0.9157$, and b) $\gamma = 2.5$ using the initial condition $x_0 = 0.7922$.	74

4.17	Cobweb diagram for the bimodal map f_D with $\gamma = \frac{3 + \sqrt{17}}{2}$ using the initial condition $x_0 = 0.0357 \in [0, 1]$.	80
4.18	Cobweb diagram for the bimodal map f_D with $\gamma = 3.6$ using the initial condition $x_0 = 0.8491 \in [0, 1]$.	82
4.19	Cobweb diagram for the bimodal map f_D with $\gamma = 4$ using the initial condition $x_0 = 0.9340 \in [0, 1]$.	83
4.20	Bifurcation diagram of the bimodal map f_D given by (4.5) with respect to the parameter $\gamma \in (0, 4] \in \mathbb{R}$.	84

Resumen.

Demostramos la existencia de la monoestabilidad y la biestabilidad en el sentido de Lyapunov en dinámicas no lineales discretas y discutimos las propiedades asociadas con este comportamiento. Específicamente, introducimos las condiciones necesarias para asegurar la ocurrencia de biestabilidad dentro de una familia paramétrica de mapeos bimodales, basados en el mapeo diferencia. El mapeo bimodal está definido dentro de una partición regular que consiste en dos subintervalos. En esta investigación, presentamos tres casos de estudio: el primer caso corresponde a mantener fijo la primera moda, mientras que la segunda moda cambia según un parámetro de bifurcación. En el segundo caso, la primera moda cambia según otro parámetro de bifurcación mientras que la segunda moda permanece fija. En el tercer caso, ambos puntos críticos asignados a las modas cambian según un parámetro de bifurcación. Se muestran diagramas de bifurcación para los tres casos de estudios y nos permiten identificar conjuntos invariantes o regiones de atrapamiento en cada subintervalo. Posteriormente, se definen dos conjuntos invariantes para habilitar biestabilidad. Por lo tanto, la monoestabilidad y biestabilidad aparecen en familias paramétricas de acuerdo con el control de conjuntos invariantes y regiones de atrapamiento al variar un parámetro de bifurcación. Los resultados numéricos se alinean con la teoría desarrollada.

Abstract.

We demonstrate the emergence of monostability and bistability in Lyapunov sense in discrete-time nonlinear dynamics and discuss the properties associated with this behavior. Specifically, we introduce the necessary conditions to ensure the occurrence of the bistability within a parametric family of bimodal maps, based on the difference map. The bimodal map is defined within a regular partition consisting of two subintervals, and we present three case studies: the first case corresponds to keeping the first modal fixed, while the second modal changes according to a parameter. In the second case, the first modal changes according to another parameter while the second modal remains fixed. In the third case, both critical points assigned to the modals change according to a bifurcation parameter. Bifurcation diagrams are shown for three case studies and let us identify invariant sets or trapping regions in each subinterval. Subsequently, two invariant sets are defined to enable the bistability. Therefore, monostability and bistability emerge in parametric families based on the control of invariant sets and trapping regions through variations in a bifurcation parameter. The numerical results are consistent with the developed theory.

Chapter 1

Introduction

In this chapter, we provide a historical overview of multimodal maps, as well as monostability and bistability, including their study and analysis within dynamical systems. We also introduce the concepts of trapping regions and invariant sets, examined in both discrete-time and continuous-time dynamical systems, such as the logistic map and the Lorenz attractor. Additionally, we discuss the motivation behind this work and outline the objective we aim to achieve. The chapter is structured into five sections: the first provides a brief overview of multimodal maps; the second delves into the background of multistability, with examples of monostable and bistable dynamical systems and important applications of multistability, as well as the concepts of trapping regions and invariant sets. The third section explains the motivation for this thesis, and the fourth section concludes with the objective of this work and an outline of the thesis.

1.1 Multimodals maps

A monoparametric multimodal map $f(x, \alpha) : I \to I$, where $I \subset \mathcal{D} \subseteq \mathbb{R}$ is a map that contains more than one maximum critical point in the interval I, which in this context are called modals and are denoted by C_i . These modals represent local maxima or minima, so if there is only one modal denoted by C_0 within I, then the map is called unimodal; if it exhibits two modals C_i in I with i = 0, 1, then it is a bimodal map. [1]. Two well-known examples of monoparametric unimodal maps are the logistic map [1]

$$f_L(x,\alpha) = \alpha x(1-x)$$
 where $\alpha \in (0,4] \subset \mathbb{R}$,

and the tent map [3]

$$f_T(x,\mu) = \begin{cases} \mu x & \text{for } x < \frac{1}{2}, \\ \mu(1-x) & \text{for } x \ge \frac{1}{2}, \end{cases} \quad \text{where} \quad \mu \in (0,2] \subset \mathbb{R},$$

since we know that both maps have a single modal in the interval $I = [0, 1] \subset \mathbb{R}$, where these modals are located at $(C_0, f_L(C_0, \alpha)) = \left(\frac{1}{2}, \frac{\alpha}{4}\right)$ for the logistic map and at $(C_0, f_T(C_0, \mu)) = \left(\frac{1}{2}, \frac{\mu}{2}\right)$ for the tent map. Figures 1.1 and 1.2 depict the logistic map and the tent map for different values of their respective parameters. Once a brief introduction of what a unimodal map is given, we will address what a multimodal map is. We know that a unimodal map has only one modal in the interval where it is defined; consequently, a multimodal map exhibits multiple modals within the defined interval. Figure 1.3 shows a family of multimodal maps. A family of maps that satisfy the brief definition given of a multimodal map are bimodal maps, which exhibit two modals within the interval where they are defined. An example of a monoparametric bimodal map is the difference map [4]

$$f_D(x,\beta) = \begin{cases} 2\beta x(1-2x) & \text{for } 0 \le x < \frac{1}{2}, \\ \\ 2\beta (x-1)(1-2x) & \text{for } \frac{1}{2} \le x \le 1. \end{cases}$$

Where $\beta \in (0, 4] \subset \mathbb{R}$. The difference map f_D is constructed by taking the difference between the logistic map and the tent map, thus creating a quadratic map that contains two unimodal maps within the interval $I = [0, 1] \subset \mathbb{R}$. Further elaboration on the aforementioned is provided in Chapter 2. The difference map serves as inspiration for the study carried out in this research work, turning the difference map into a biparametric map (dependent on parameters β_1 and β_2). Figure 1.4 illustrates the difference map.



Figure 1.1: The logistic map for different values of the parameter α .



Figure 1.2: The Tent map for different values of the parameter μ .



Figure 1.3: Family of multimodal maps based on the logistic map [1].



Figure 1.4: Difference map showing different values of the parameter β .

After providing a brief explanation of multimodal maps, we will gain a clearer understanding of what constitutes a family of monoparametric and biparametric bimodal maps. This concept is crucial because, within this family, we will investigate the necessary conditions for achieving monostability and bistability.

1.2 Monostability, bistability, trapping region and invariant set in dynamical systems

Stability can have different meanings and definitions in the field of dynamical systems, but in this chapter, our goal is to provide a clear interpretation. We use the term *stable* (specifically *monostable* in this context) to describe a system that maintains a consistent behavior within a defined set of boundaries over time, showing only one stable state. If the same system exhibits this behavior within two different sets of boundaries, it is termed *bistable*, indicating two stable states. Generally, if a system exhibits this behavior across multiple separate sets of boundaries, it is called *multistable*, meaning it has multiple stable states.

To explain further, a *stable* discrete dynamical system is one where initial conditions lead to orbits—sequences of points that describe the system's evolution as time progresses—that converge towards a fixed point. When orbits converge towards a specific region instead of a single fixed point, we refer to it as *generalized stability*. In monostable systems, all orbits from initial conditions converge towards a single fixed point. In bistable systems, orbits from different initial conditions converge towards a single fixed point. In bistable systems, orbits from different initial conditions converge towards one of two possible fixed points. Similarly, in multistable systems, orbits can converge towards multiple distinct fixed points.

The term multistability was first introduced in 1971 in the context of a psychological study on visual perception [2]. This study explored how a single image could be interpreted in fundamentally different ways depending on the viewer's initial visual focus. It demonstrated that the same visual input could lead to various distinct perceptions, each representing a different stable state. This observation gave rise to the concept of multistability, which suggests that an image—or more broadly, a system—can exhibit multiple stable states or attractors, each of which can be perceived or realized depending on initial conditions or contextual factors. In essence, multistability reflects the idea that certain visual stimuli or dynamic systems can possess several equally



Figure 1.5: Image first presented in 1915 with the title 'My Wife and My Mother-in-law', where the bistability in visual perception is demonstrated [2].

valid and stable interpretations or states, depending on how they are observed or interacted with. In the field of dynamical systems, in brief, a multistable system has the ability to have multiple stable states [5]. If we relate visual multistability to dynamical systems, we say that the initial visual focus is analogous to the initial condition, which, depending on where we place our initial visual focus, will be the perception we have of a multistable image, where each possible perception is a stable state.

In Figure 1.5, we observe a bistable image, a classic example of multistability in visual perception. This type of image illustrates how the same visual stimulus can be interpreted in multiple stable ways. Specifically, if we focus our attention on point c in the image, the figure that appears is that of a young woman. This interpretation is based on the characteristics of the face and profile that are configured when focusing on point c. On the other hand, when we direct our attention to point a, the image reveals the figure of an elderly woman. In this perspective, the facial features are rearranged in such a way that the image is perceived as the face of an older woman.



Figure 1.6: Bistability in the energetic description of a system.

In this work, as mentioned earlier, the characteristics of interest are monostability and bistability. In Figure 1.6, we can observe the energy graph of a system that exhibits two stable states (bistability), in which three equilibrium points are denoted by colored spheres and marked with the numbers 1, 2, and 3. Let us observe that equilibrium points 1 and 3 are stable, while equilibrium point 2 is unstable. Note that if we consider only the interval $X_1 \leq X < X_2$, the system is monostable, similarly considering $X_2 < X \leq X_3$. Additionally, note that the equilibrium point X_2 is both the upper and lower bound in these defined intervals. With this, it is evident that under any arbitrary initial condition, regardless of how close it is to the unstable equilibrium point, the trajectory will converge to the respective stable equilibrium point depending on the interval in which the initial condition lies.

The logistic map $f_L(x, \alpha) = \alpha x(1-x)$ is clearly an example of a monostable system, exhibiting two equilibrium points (fixed points in discrete time) in the interval $I = [0, 1] \subset \mathbb{R}$, which are x = 0 and $x = \frac{\alpha - 1}{\alpha}$. We say that the logistic map is a monostable system because for $0 < \alpha \leq 4$, the dynamics converge to a fixed point of f_L or remain within the interval I. It is observed that when the parameter $0 < \alpha < 1$, the point x = 0 is the only stable point in the interval $I = [0, 1] \subset \mathbb{R}$. However, when $1 < \alpha < 3$, the origin changes



Figure 1.7: Logistic map for different values of $\alpha \in (0, 4] \subset \mathbb{R}$: (a) $\alpha = 3.8$ and (b) $\alpha = 4$.

from stable to unstable, and the point $x = \frac{\alpha - 1}{\alpha}$ becomes stable. However, when $3 < \alpha \leq 4$, both equilibrium points are unstable. Nonetheless, if we wish to study when the dynamics diverge, that is, when the dynamics no longer remain within the interval I, we can observe that for $\alpha > 4$, the dynamics escape from the interval I after a certain number of iterations. This is because the modal of f_L is located at $(C_0, f_L(C_0, \alpha)) = \left(\frac{1}{2}, \frac{\alpha}{4}\right)$, so we observe that for $\alpha > 4$, the modal does not belong to the interval I, since $\frac{\alpha}{4} > 1$. Therefore, when $\alpha > 4$ the logistic map is not a monostable system. However, two important concepts arise here, which are trapping region and invariant set.

Let us observe in Figure 1.7a the behavior of the logistic map when the parameter $\alpha = 3.8$, while in Figure 1.7b when $\alpha = 4$, both for initial conditions within I. We can observe from Figure 1.7a that when $\alpha = 3.8$, it holds that $f_L(I) \subset I$ since it does not cover the entire interval I, whereas when $\alpha = 4$, we satisfy $f_L(I) = I$ as we see that the trajectory covers the entire interval I. With this, two fundamental concepts for the study of monostability and bistability in a family of bimodal maps are introduced, namely trapping region and invariant set. Thus, briefly, we can observe that when $0 < \alpha < 4$, there exists a trapping region since $f_L(I) \subset I$, and when $\alpha = 4$, there is an invariant set, as $f_L(I) = I$ is fulfilled. The precise definitions of trapping region and invariant set are provided in Chapter 2. We will review some applications and studies of the multistability in dynamic systems of science and technology. As mentioned earlier, multistability appears in different dynamic systems within science and technology. For example, in Biology, genetic switches have been constructed in biological systems that exhibit multistability [6]. In physics, the study of multistability in networks of oscillators [7], and within Chemistry, the bi-stability and multi-stability occurring within an ethanol reactor have been studied using numerical tools [8]. Some other studies include the investigation of multistability in symmetric chaotic systems [9], multistability of self-reproducing chaotic systems [5], the study of multistability in a discrete economic model [10], multistability in a memristive chaotic system [11], the multistability in social systems [12], among others [13].

The bistability is essential in neuronal dynamics, as it is present in neuronal interactions as well as in cellular signaling [14]. Within climatological dynamics, coexisting attractors related to bistability have been found through deep oceanic convection [15, 16]. Similarly, in electronics, multivibrators—classified as astable, monostable, and bistable—play a crucial role. An astable multivibrator alternates continuously between two states, generating a square wave without an external input signal, and is commonly used in clock generators and oscillators [17]. A monostable multivibrator has one stable and one unstable state; upon receiving a trigger, it switches to the unstable state for a period before returning to the stable state, making it useful for timing applications such as pulse generators [17]. A bistable multivibrator, or flip-flop, has two stable states and remains in one state until an external signal switches it to the other. This type is fundamental in digital memory construction and data storage registers [18]. These configurations are widely used in digital and analog electronics for various synchronization, storage, and signal generation functions.

1.3 Motivation

The dynamics of a family of multimodal maps with non-uniform intervals have been previously studied [19], where it was found that a bifurcation parameter could induce changes in the dynamics of this family. That work serves as motivation, as multistability was numerically observed; however, the study of multistability was not the primary objective of that research. Inspired by the multistability observed in this previous work, this investigation focuses specifically on monostability and bistability within a family of parametric bimodal maps with uniform intervals, from both theoretical and numerical perspectives.

This work takes as its starting point the investigation of a family of bimodal maps with uniform intervals, with the purpose of approaching this work from a less complex standpoint, as well as becoming familiar with the study of multimodal maps. Subsequently, once the considerations for the existence of bistability in a family of bimodal maps are established, the conditions for the existence of monostability within this map are provided. Later, to verify the theoretical study, the relevant simulations are carried out to demonstrate a relationship between the numerical and theoretical results.

1.4 Objective

The general objective of this thesis is to study the necessary conditions to guarantee the existence of monostability and bistability in a parametric family of bimodal maps. Specific objectives include constructing a series of propositions to ensure monostability and bistability, followed by conducting numerical simulations based on these propositions to verify and observe bistability. Another specific objective is to characterize trapping regions, as well as invariant sets and points within this family of maps.

1.5 Organization

This thesis is divided into chapters. Chapter 1 provides a brief overview of multimodal maps, including a historical review of multistability. A dynamical system is presented to illustrate monostability, as well as to define invariant sets and trapping regions. Finally, the applications of monostability and bistability in science and technology are discussed. The chapter concludes with an explanation of the motivation behind this work and the general and specific objectives of the thesis.

Chapter 2 introduces the definitions and theorems that form the foundation for demonstrating the obtained results and enriching the theoretical development. Additionally, this chapter defines the parametric bimodal map based on the difference map, which will be used to study bistability. Chapter 3 presents the analytical results, including a series of propositions that ensure bistability in the proposed bimodal map from Chapter 2.

Chapter 4 presents three case studies based on the proposed bimodal map. For each case, the relevant analytical and numerical results are provided using the findings from Chapter 3, aiming to demonstrate the relationship between theoretical and numerical results and to observe monostability and bistability. Chapter 4 concludes with the presentation of bifurcation diagrams for each case study. Finally, Chapter 5 presents the conclusions and outlines directions for future work.

Chapter 2

Mathematical preliminaries

In this chapter, important mathematical foundations that appear throughout the development of this research are presented. The content presented in this chapter consists of definitions and theorems within the field of real number topology, linear algebra, mathematical analysis, and discrete dynamical systems. Additionally, definitions of unimodal and bimodal maps are provided in this chapter, along with examples of these aforementioned maps, concluding with the bimodal map used for the development of this research.

2.1 Topology of the Real Numbers

Below are some definitions found in the field of real number topology. These definitions will help us better understand concepts presented in this chapter, as well as the analytical results obtained. The definitions and theorems within this section include: the existence of a norm, interior point, open set, closed set, bounded set, compact set, and metric space [20].

Definition 1 Let $x = (x_1, x_2, \dots, x_{n-1}, x_n)$ and $y = (y_1, y_2, \dots, y_{n-1}, y_n)$ be in \mathbb{R}^n . We define;

a) Equality:

x = y if, and only if, $x_1 = y_1, x_2 = y_2, \cdots, x_{n-1} = y_{n-1}, x_n = y_n$.

b) Sum:

$$x + y = (x_1 + y_1, x_2 + y_2, \cdots, x_{n-1} + y_{n-1}, x_n + y_n).$$

c) Multiplication by real numbers (scalars):

$$a\mathbf{x} = (ax_1, ax_2, \cdots, ax_{n-1}, ax_n) \quad a \in \mathbb{R}.$$

d) Difference:

$$\boldsymbol{x} - \boldsymbol{y} = \boldsymbol{x} + (-1)\boldsymbol{y}.$$

e) Zero vector or origin:

$$\theta = (0, 0, \cdots, 0, 0).$$

f) Inner product or dot product:

$$\boldsymbol{x} \cdot \boldsymbol{y} = \sum_{k=1}^{n} x_k y_k$$

g) Norm o length

$$\|\boldsymbol{x}\| = (\boldsymbol{x} \cdot \boldsymbol{x})^{1/2} = \left(\sum_{k=1}^{n} x_k^2\right)^{1/2}$$

Theorem 1 Let x and y be points in \mathbb{R}^n , then

- 1. $||x|| \ge 0$
- 2. $||a\mathbf{x}|| = |a|||\mathbf{x}|| \quad \forall a \in \mathbb{R}$
- 3. $\|\boldsymbol{x} \cdot \boldsymbol{y}\| \leq \|\boldsymbol{x}\| \|\boldsymbol{y}\|$ Cauchy-Schwarz inequality
- 4. $\|\mathbf{x} + \mathbf{y}\| \le \|\mathbf{x}\| + \|\mathbf{y}\|$ Triangle inequality.

With the existence of a norm in the space \mathbb{R}^n , we can provide the definition of an *n*-ball. This definition is presented in order to enunciate and define in a clearer way the concepts of sets presented later.

Definition 2 Let a be a point in \mathbb{R}^n and let $r \in \mathbb{R}^+$. The set of all points x in \mathbb{R}^n such that

$$||x - a|| < r$$

is called an open n-ball of radius r with center at \mathbf{a} , usually denoted as $\mathbf{B}(\mathbf{a}, r)$ or $\mathbf{B}(\mathbf{a})$ [20].

Definition 3 Let $S \subset \mathbb{R}^n$ and $a \in S$. Then a is called an interior point of S if there exists a B(a) where all its points belong to S [20].

From Definition 2, we observe that we have an *n*-ball whose boundary is not contained since the distance between any point x contained in the ball (different from its center) and its center x must be strictly less than its radius r. In Definition 3, an interior point is defined, where this point is interior if it belongs to a set S and there exists an $\varepsilon > 0$ such that the *n*-ball centered at a with radius ε is completely contained in S. In mathematical context, it is expressed as

 $\exists \varepsilon > 0 \text{ such that } B(x, \varepsilon) \subseteq S.$

Note that the radius r presented in Definition 2 is replaced by ε , however, the interpretation is the same for both cases. Next, the definition for an open set is presented, this definition is given since following Definition 4, using the definition of an open set, what a closed set is defined.

Definition 4 A set $S \in \mathbb{R}^n$ is called open if all its points are interior points [20].

Definition 5 Let $S \subset \mathbb{R}^n$ be a set, S is said to be closed if its complement is an open set [20].

If we observe, the previous definition mentions a set operation called complement, let us remember that this operation generates a new set, usually denoted as $S^c \subset \mathbb{R}^n$ containing the elements that do not belong to the set S, mathematically it is expressed as

$$\mathcal{S}^c = \{ x \in \mathbb{R}^n : x \notin \mathcal{S} \}$$

so that if we take the union of these two sets

 $\mathbb{S}^{c}\cup\mathbb{S}=\mathbb{R}^{n}$

and according to the definition of complement, we know that the elements not contained in the set S necessarily have to be in S^c , therefore, S^c has to be a closed set in order to generate the entire space \mathbb{R}^n where these sets are defined. Despite the previous definition, an alternative definition of a closed set will be presented later, however, to reach this definition, it is necessary to present the definition of a limit point.

Definition 6 If $S \subseteq \mathbb{R}^n$ and x is a point in \mathbb{R}^n where x is not necessarily contained in S. Then x is a limit point of S if every n-ball B(x) contains at least one point of S [20].

A limit point is also known as a limit or accumulation point. In simple words, a point contained in S is considered a limit point if, no matter how small the *n*-ball is, it will always be possible to find at least another point belonging to S and within the *n*-ball. Knowing what a limit point is, we proceed to provide the definition of a closed set.

Definition 7 A set $S \in \mathbb{R}^n$ is closed if and only if it contains all its limit points [20].

Definition 8 A set $S \in \mathbb{R}^n$ is said to be bounded if it is completely contained within an n-ball B(a, r) for some r > 0 and some $a \in \mathbb{R}^n$ [20].

Definition 9 A set $S \in \mathbb{R}^n$ is compact if it is closed and bounded [21].

Definition 10 A set \mathfrak{X} is said to be a metric space if to every two points $p, q \in \mathfrak{X}$ there is associated a real number d(p,q) called distance from p to q, such that

- 1. d(p,q) > 0 if $p \neq q$; d(p,p) = 0;
- 2. d(p,q) = d(q,p);
- 3. $d(p,q) \leq d(p,r) + d(r,q)$, for all $r \in \mathfrak{X}$.

2.2 Linear Algebra

In this section, some concepts used in Linear Algebra are defined. The presented definitions are; field, vector space, and linear transformation. These definitions are presented in order to enunciate what a linear transformation is since when defining the derivative of a function, it is mentioned that the derivative is a linear transformation.

Definition 11 A field \mathcal{F} is a set with two operations + and \cdot called addition and multiplication respectively, they are defined such that, for each pair of elements $x, y \in \mathcal{F}$ there exist unique elements x + y and $x \cdot y$ in \mathcal{F} for which the following conditions are satisfied. Let $a, b, c \in \mathcal{F}$

- 1. a + b = b + a and $a \cdot b = b \cdot a$ (commutativity in addition and multiplication).
- 2. (a+b)+c = a+(b+c) and $(a \cdot b) \cdot c = a \cdot (b \cdot c)$ (associativity in addition and multiplication).
- 3. There exist distinct elements 0 and 1 in \mathcal{F} such that

$$0 + a = a \qquad 1 \cdot a = a$$

(existence of identity elements in addition and multiplication).

4. For each element $a \in \mathcal{F}$ and for each element $b \in \mathcal{F}$ with $b \neq 0$, there exist elements $c, d \in \mathcal{F}$ such that

$$a+c=0$$
 $b\cdot d=1$

(existence of inverse for addition and multiplication).

5. $a \cdot (b + c) = a \cdot b + a \cdot c$ (distributivity of multiplication over addition) [22].

Definition 12 A vector space \mathcal{V} over a field \mathcal{F} consists of a set in which two operations are defined (called addition and multiplication), such that for any $x, y \in \mathcal{V}$ there exists a unique x + y in \mathcal{V} , and for each $a \in \mathcal{F}$ and $x \in \mathcal{V}$ there exists a unique element $ax \in \mathcal{V}$, so that the following conditions are met. Let $x, y, z \in \mathcal{V}$

- 1. x + y = y + x (commutativity in addition).
- 2. (x+y) + z = x + (y+z) (associativity in addition).
- 3. There exists an element in \mathcal{V} denoted by 0 such that x + 0 = x for any element $x \in \mathcal{V}$ (additive identity).
- 4. For each element $x \in \mathcal{V}$ there exists a unique element $y \in \mathcal{V}$ such that x + y = 0 (additive inverse).
- 5. For each element $x \in \mathcal{V}$, 1x = x (multiplicative identity).
- 6. For each pair of elements $a, b \in \mathcal{F}$ it holds that (ab)x = a(bx) (associativity in multiplication).
- 7. For each element $a \in \mathcal{F}$ it holds that a(x + y) = ax + ay (distributive property).
- 8. For each pair of elements $a, b \in \mathcal{F}$ it holds that (a + b)x = ax + bx(distributive property) [22].

Definition 13 Let \mathcal{V} and \mathcal{W} be vector spaces over \mathcal{F} . We call a function $T: \mathcal{V} \to \mathcal{W}$ a linear transformation from \mathcal{V} to \mathcal{W} if, $\forall x, y \in \mathcal{V}$ and $c \in \mathcal{F}$, the following conditions are met:

- 1. T(x+y) = T(x) + T(y).
- 2. T(cx) = cT(x).

2.3 Mathematical Analysis

In the development presented in the following Chapter, functions with a set of specific characteristics are worked with, among these are continuity and differentiability. Therefore, the definitions of function are presented, the limit operator of a function and its existence are defined, furthermore, once the limit operator is defined, continuity and derivative of a function are defined. **Definition 14** A function f is a rule that assigns each element of a set \mathfrak{D} to a unique element of a second set \mathfrak{C} . The sets \mathfrak{D} and \mathfrak{C} are called the domain and codomain of the function. The set of elements of the codomain \mathfrak{C} that have an element of the set \mathfrak{D} assigned to them is called the range of the function. The notation $f: \mathfrak{D} \to \mathfrak{C}$ is used to indicate a function f with domain \mathfrak{D} and codomain \mathfrak{C} . If we use $f: \mathfrak{D} \to \mathfrak{D}$, it indicates that the domain and codomain are the same set [23].

Definition 15 Let $f : \mathcal{D} \to \mathbb{R}$ be a function, $I \subseteq \mathcal{D} \subseteq \mathbb{R}$ an open interval and let $x \in I$ be a point of I. Then $L \in \mathbb{R}$ is the limit of a function f at xdenoted as

$$\lim_{z \to x} f(z) = L$$

if and only if for every $\varepsilon > 0$ there exists a $\delta > 0$ such that $|z - x| < \delta$, then we have $|f(z) - L| < \varepsilon$ [24, 25].

Definition 16 Let $\mathcal{D} \subseteq \mathbb{R}$ be an open interval of nonzero length from which at most a finite number of points have been removed, and let $f : \mathcal{D} \to \mathbb{R}$ be a function. Then f is called continuous if and only if

- 1. f(x) is defined, i.e., $x \in \mathcal{D}$, and
- 2. $\lim_{z\to x} f(z)$ exists, and
- 3. $\lim_{z\to x} f(z) = f(x)$, and
- 4. if x is an endpoint of \mathcal{D} , use the left or right limits in 2 and 3, as appropriate.

f is called continuous on \mathcal{D} if and only if f is continuous at each $x \in \mathcal{D}$ [24].

Definition 17 Let $f : \mathbb{R} \to \mathbb{R}$ be a function and let $x \in \mathbb{R}$ be a point. f is said to be differentiable at x if there exists a linear transformation $Df(x) \in \mathbb{R}$ that satisfies

$$\lim_{|h| \to 0} \frac{|f(x+h) - f(x) - Df(x)h|}{|h|} = 0$$

The linear transformation Df(x) is called the derivative of f at x [26].

It is worth noting that the previous definition may confuse the reader since, as such, the usual definition of derivative is not provided. However, it should be noticed that for the limit presented in Definition 17 to be zero, it is necessary that the linear transformation Df(x) has the following form

$$Df(x) = \lim_{|h| \to 0} \frac{|f(x+h) - f(x)|}{|h|}$$

which is the definition that is usually presented when starting with the concept of derivative.

Definition 18 Let $f : \mathcal{D} \to \mathbb{R}$ be a function. Suppose f is differentiable on \mathcal{D} . Then $f \in C^1$ if the derivative $Df : \mathcal{D} \to \mathbb{R}$ is continuous on \mathcal{D} [26].

2.4 Discrete-time Dynamical Systems

In this section, definitions within the area of dynamical systems are presented, which involve the definitions of discrete dynamical system, fixed point, periodic point, and orbit. Subsequently, the conditions for a fixed point to be an attractor or a repeller are presented, as well as the conditions for an orbit to be attractive or repulsive. Finally, the definition of chaos in the sense of Devaney, multimodal map, and some examples of unimodal and bimodal maps are given.

Definition 19 A discrete dynamical system is a relation of the form

$$x_{k+1} = f(x_k), \quad k \in 0, 1, 2, .., N$$

$$(2.1)$$

where $x_k \in \mathbb{R}$ and x_0 is the initial condition [27].

Definition 20 Let $f : I \to I$ be a function and let $c \in I$ be a point. Then, if it satisfies that f(c) = c, then c is a fixed point of f [23].

Theorem 2 Let I = [a, b] be a closed interval and let $f : I \to I$ be a continuous function. Then f has a fixed point in I [23].

Definition 21 Let f be a function. Then x is a periodic point of f with period k if $f^k(x) = x$. In other words, a point is a periodic point of f with period k if it is a fixed point of f^k . The periodic point x has primary period k_0 if $f^{k_0}(x) = x$ and $f^n(x) \neq x$ when $0 < n < k_0$ [23, 28]. Where

$$f^{2}(x) = (f \circ f)(x)$$

$$f^{3}(x) = (f \circ f \circ f)(x) = (f \circ f^{2})(x)$$

$$f^{4}(x) = (f \circ f \circ f \circ f)(x) = (f \circ f^{3})(x)$$

$$\vdots$$

$$f^{k}(x) = (\underbrace{f \circ f \cdots f \circ f}_{k-times})(x) = (f \circ f^{k-1})(x) \quad with \ k \in \mathbb{N}$$

Definition 22 Let $f : I \to I$ be a function and $x_0 \in I$ a point, we define the orbit of x_0 under f, denoted as $\mathcal{O}(x_0, f)$, as the sequence of points

$$\mathcal{O}(x_0, f) = x_0, f(x_0), f^2(x_0), \cdots f^{k-1}(x_0), f^k(x_0)$$

The point x_0 is called the initial condition of the orbit [29].

Definition 23 Let $x^* \in I$ be the fixed point of the discrete-time dynamical system defined by the equation $x_{k+1} = f(x_k)$ denoted by (2.1) Then;

- the fixed point x^* is stable in the Lyapunov sense if for each $\varepsilon > 0$ there exists $\delta = \delta(\varepsilon)$ such that for any $x \in I$ with $|x - x^*| < \delta$, then $|f^n(x^*) - f^n(x)| < \varepsilon$, for all $n \in \mathbb{N}$ large enough.
- The fixed point x^* is unstable if it is not stable.
- the fixed point x^* is asymptotically stable in the Lyapunov sense if it is stable in the Lyapunov sense and also if there exists $\delta = \delta(\varepsilon)$ such that for any $x \in I$ with $|x - x^*| < \delta$, then $\lim_{n \to \infty} |f^n(x^*) - f^n(x)| = 0$ [27, 30].

Definition 24 Let $f : I \to I$ be a C^1 function and let $x \in I$ be a fixed point of f. Then x is an attracting fixed point if |Df(x)| < 1. The point x is a repelling fixed point if |Df(x)| > 1. Finally, the fixed point x is a neutral fixed point if |Df(x)| = 1 [29]. **Definition 25** Let $f : I \to I$ be a C^1 function and let x_1, x_2, \dots, x_k be a periodic orbit of period k, where $x_i \in I$ for all $i \in 1, 2, \dots, k$. Then, the periodic orbit is attracting if [31];

$$|Df(x_1) \cdot Df(x_2) \cdots Df(x_k)| < 1$$

and repelling if

$$|Df(x_1) \cdot Df(x_2) \cdots Df(x_k)| > 1$$

Definition 26 Suppose \mathfrak{X} is a set and $\mathfrak{Y} \subset \mathfrak{X}$. We say that \mathfrak{Y} is dense in \mathfrak{X} if, for every point $x \in \mathfrak{X}$, there exists a point $y \in \mathfrak{Y}$ arbitrarily close to x [29].

Definition 27 A dynamical system f is transitive if for every pair of points x, y and any $\varepsilon > 0$ there exists a third point z within ε of x whose orbit is within ε of y [29].

Definition 28 A dynamical system f exhibits sensitive dependence on initial conditions if there exists a $\beta > 0$ such that, for any x and any $\varepsilon > 0$, there exists $y \in \mathbb{R}^+$ within ε of x and $k \in \mathbb{N}$ such that the distance between $f^k(x)$ and $f^k(y)$ is at least β [29].

Definition 29 A dynamical system f is chaotic if;

- 1. The periodic points of f are dense.
- 2. f is transitive.
- 3. f exhibits sensitive dependence on initial conditions [29].

Definition 30 Let I = [0,1] be a compact set, the mapping $f: I \to I$ is called k-modal if it is continuous on I and moreover, it has k critical points denoted by c_i with $i \in 0, 1, ..., k - 1$ in I. Furthermore, there exist intervals I_i such that $\bigcup_{i=1}^k I_{i-1} = I$ such that $\forall i = 0, 1, ..., k - 1$ it holds that $c_i \in I_i$ and $f(c_i) > f(x, \gamma, \beta_i) \ \forall x \in I_i$ and $x_i \neq c_i$, where γ, β_i are parameters. The case when it has only one critical point, i.e., k = 1, the mapping is called unimodal, while the case when it has two critical points (k = 2) is called bimodal [4]. The logistic map and the tent map are examples of unimodal maps, as both are continuous on I with a single critical point, moreover, they are increasing for $x \in \left[0, \frac{1}{2}\right)$ and decreasing for $x \in \left(\frac{1}{2}, 1\right]$ [4]. Additionally, in Chapter 1, Section 1.1, we discuss and analyze the monostability of the logistic and the tent map. The definitions of these maps are provided below.

Definition 31 Let $I = [0,1] \subset \mathbb{R}$. We refer to $f_L : I \to I$ as the logistic map defined as

$$f_L(x,\alpha) = \alpha x(1-x), \qquad (2.2)$$

where the parameter $\alpha \in (0, 4] \subset \mathbb{R}$ [4].

Definition 32 Let $I = [0, 1] \subset \mathbb{R}$. We refer to $f_T : I \to I$ as the tent map, expressed as

$$f_T(x,\mu) = \begin{cases} \mu x & \text{for } x < \frac{1}{2}, \\ \mu(1-x) & \text{for } x \ge \frac{1}{2}, \end{cases}$$
(2.3)

where $\mu \in (0, 2]$ is a parameter [4].

The bimodal map used for this work is based on the difference map $f_D(x,\beta)$, which is introduced in [4].

Definition 33 Let f_L and f_T be the logistic map and the tent map shown in (2.2) and (2.3) respectively, taking the maximum value of their parameters $\alpha = 4$ and $\mu = 2$. We define $f_D(x,\beta)$ as the difference between the logistic map and the tent map multiplied by the parameter $\beta \in (0,4]$, i.e., $f_D(x,\beta) = \beta(f_L(x,4) - f_T(x,2))$, as follows:

$$f_D(x,\beta) = \begin{cases} 2\beta x(1-2x) & \text{for } 0 \le x < \frac{1}{2}, \\ 2\beta (x-1)(1-2x) & \text{for } \frac{1}{2} \le x \le 1. \end{cases}$$
(2.4)
We can observe that the difference map $f_D(x,\beta)$ is continuous on I and has two critical points located at $c_0 = (x, f_D(x,\beta)) = \left(\frac{1}{4}, \frac{\beta}{4}\right)$ and $c_1 = (x, f_D(x,\beta)) = \left(\frac{3}{4}, \frac{\beta}{4}\right)$ and $f_D(c_0,\beta) = f_D(c_1,\beta) = \frac{\beta}{4}$.

Definition 34 A subset $\mathcal{A} \subset I$ is said to be an invariant set for a discretetime dynamical system (2.1) if it satisfies that $f^n(\mathcal{A}) = \mathcal{A}, \forall; n \in \mathbb{N}$ [32].

Definition 35 A subset $\mathcal{A} \subset I$ is called a trapping region of a discrete-time dynamical system (2.1) if for every orbit of $x_0 \in \mathcal{A}$ will move to the region's interior and remain there as the system evolves $f^n(\mathcal{A}) \subset int(\mathcal{A}), \forall; n \in \mathbb{N}$ [32].

Definition 36 A family of functions f_{β} undergoes a saddle-node (or tangent) bifurcation at the parameter value $\beta = \beta_0$ if there exists an open interval I and an $\varepsilon > 0$ such that:

- 1. For $\beta_0 \varepsilon < \beta < \beta_0$, f_β has no fixed points in the interval I.
- 2. For $\beta = \beta_0$, f_β has one fixed point in I, and this fixed point is neutral.
- 3. For $\beta_0 < \beta < \beta_0 + \varepsilon$, f_β has two fixed points in I, one attracting and one repelling [29].

Definition 37 A family of functions f_{β} undergoes a period-doubling bifurcation at the parameter value $\beta = \beta_0$ if there exists an open interval I and an $\varepsilon > 0$ such that:

- 1. For each β in the interval $[\beta_0 \varepsilon, \beta_0 + \varepsilon]$, there exists a unique fixed point p_β for f_β in I.
- 2. For $\beta_0 \varepsilon < \beta \leq \beta_0$, f_β does not have period-2 cycles in I, and p_β is an attractor.
- 3. For $\beta_0 < \beta < \beta_0 + \varepsilon$, exists a unique period-2 cycle q_{β}^1 , q_{β}^2 in I with $f_{\beta}(q_{\beta}^1) = q_{\beta}^2$. This period-2 cycle is attractive, while the fixed point p_{β} is repelling.

4. As
$$\beta \to \beta_0$$
, we have $q^i_\beta \to p_{\beta_0}$, $\forall i = \{1, 2\}$ [29].

In the next chapter, the parametric family of bimodal maps that will be used is introduced and defined. Additionally, a series of theoretical results are presented that ensure bistability within this family. These results include an analysis of fixed points, saddle-node and period-doubling bifurcations, as well as the characterization of invariant sets and trapping regions that give rise to bistability.

Chapter 3

Bistability in a bimodal map

The family of bimodal maps based on the difference map (2.4) that will be used for this study is defined as follows:

$$f_D(x,\gamma,\beta_1,\beta_2) = \begin{cases} f_{D_1} & \text{for } \zeta_0 \le x < \zeta_1, \\ f_{D_2} & \text{for } \zeta_1 \le x \le \zeta_2. \end{cases}$$
(3.1)

Where f_{D_1} and f_{D_2} are unimodal maps defined as:

$$f_{D_1}(x,\gamma,\beta_1) = \gamma \beta_1(x-\zeta_0)(\zeta_1-x),$$
(3.2)

$$f_{D_2}(x,\gamma,\beta_2) = \gamma \beta_2(x-\zeta_1)(\zeta_2-x),$$
(3.3)

and the parameters $\gamma, \beta_i \in (0, 4] \subset \mathbb{R}$ must satisfy that:

$$0 < \gamma \beta_i \le \frac{4}{(\zeta_i - \zeta_{i-1})^2},\tag{3.4}$$

for i = 1, 2 and $\zeta_k \in [0, 1] \subset \mathbb{R}$ for k = 0, 1, 2 where it holds that $\zeta_0 < \zeta_1 < \zeta_2$.

The bistability arises when a dynamic system presents two invariant sets, for the same parameter values, i.e. for a dynamical system $f: I \to I$ there are two invariant sets \mathcal{A}_1 and \mathcal{A}_2 such that $f(\mathcal{A}_1) \subset \mathcal{A}_1$ and $f(\mathcal{A}_2) \subset \mathcal{A}_2$, with $\mathcal{A}_1 \subset I$ and $\mathcal{A}_2 \subset I$. In this Chapter, we show that the dynamical system (3.1) presents two invariant sets in the interval I = [0, 1] for a set of parameter values and the system oscillates in one of them depending of the initial condition x_0 . The analysis of fixed points and their stability are of great importance to understand the bifurcations that a dynamical system presents. Therefore we will begin to study the fixed points of the system (3.1).

Proposition 1 The bimodal map defined in (3.1) presents fixed points in the interval $\zeta_1 \leq x \leq \zeta_2$ if:

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.5)

Proof. For find the upper bound of the interval where the parameter β_2 take it values with the purpose that the map f_D has fixed points in the interval $[\zeta_1, \zeta_2] \subset (0, 1] \subset \mathbb{R}$ it is suffices to find the maximum point that the second modal can take, which, it is located at the point $(x, f(x)) = \left(\frac{\zeta_1 + \zeta_2}{2}, 1\right)$, knowing the maximum point that f_D can take, we can substitute into the equation $f_{D_2}(x, \gamma, \beta_1)$ showed in (3.3), thus obtaining

$$f_{D_2}\left(\frac{\zeta_1+\zeta_2}{2},\gamma,\beta_1\right) \le 1,$$

analogously, we have the following expression

$$\gamma\beta_2\left(\frac{\zeta_1+\zeta_2}{2}-\zeta_1\right)\left(\zeta_2-\frac{\zeta_1+\zeta_2}{2}\right)\leq 1,$$

by simplifying, we obtain

$$\gamma\beta_2\left(\frac{\zeta_1+\zeta_2-2\zeta_1}{2}\right)\left(\frac{2\zeta_2-\zeta_1-\zeta_2}{2}\right) = \gamma\beta_2\left(\frac{\zeta_2-\zeta_1}{2}\right)\left(\frac{\zeta_2-\zeta_1}{2}\right) \leq 1,$$

again, by simplifying and solving the preceding inequality for the parameter β_2 , we obtain the maximum bound

$$\beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2},\tag{3.6}$$

now, for find the lower bound of the interval $\zeta_1 \leq x \leq \zeta_2$, we proceed to find the fixed points for the second modal showed by (3.5), thus, to located the fixed points, we perform $f_{D_2}(x, \gamma, \beta_2) = x$, thus

$$\gamma\beta_2(x-\zeta_1)(\zeta_2-x)=x,$$

by expanding the preceding equation, thus obtaining

$$\gamma\beta_2 x^2 - \gamma\beta_2(\zeta_1 + \zeta_2)x + \gamma\beta_2\zeta_1\zeta_2 = x_2$$

by simplifying and setting equal to zero, thus

$$\gamma \beta_2 x^2 - [\gamma \beta_2 (\zeta_1 + \zeta_2) - 1] x + \gamma \beta_2 \zeta_1 \zeta_2 = 0, \qquad (3.7)$$

by solving (3.7), the expression for the fixed point is obtained

$$x = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) - 1 \pm \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}}{2\gamma \beta_2},$$
 (3.8)

the preceding equation involves solutions in the complex numbers and the real numbers field, however, the difference map described in (3.1) that maps into the real numbers field, hence the discriminant Δ of (3.8) must satisfy that $\Delta > 0$ since we desire that f_{D_2} has two fixed points, thus

$$\Delta = [\gamma \beta_2 (\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2,$$

with the above, the inequality holds

$$[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2 - 4\gamma^2\beta_2^2\zeta_1\zeta_2 > 0, \qquad (3.9)$$

by expanding the preceding inequality, and use the parameter β_2 as variable, is obtained

$$\gamma^2 \beta_2^2 (\zeta_1 + \zeta_2)^2 - 2\gamma \beta_2 (\zeta_1 + \zeta_2) + 1 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2 > 0,$$

grouping terms, we obtain

$$[\gamma^2(\zeta_1 + \zeta_2)^2 - 4\gamma^2\zeta_1\zeta_2]\beta_2^2 - 2\gamma(\zeta_1 + \zeta_2)\beta_2 + 1 > 0, \qquad (3.10)$$

by solving the preceding expression by the general formula, is obtained

$$\beta_2 = \frac{\gamma(\zeta_1 + \zeta_2) \pm \sqrt{[\gamma(\zeta_1 + \zeta_2)]^2 - [\gamma^2(\zeta_1 + \zeta_2)^2 - 4\gamma^2\zeta_1\zeta_2]}}{\gamma^2(\zeta_2 - \zeta_1)^2}, \qquad (3.11)$$

by simplifying the terms within the square root

$$\beta_2 = \frac{\gamma(\zeta_1 + \zeta_2) \pm \sqrt{4\gamma^2 \zeta_1 \zeta_2}}{\gamma^2 (\zeta_2 - \zeta_1)^2},$$

by simplifying the preceding equation, we obtain

$$\beta_2 = \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} \pm \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2,$$

finally, by the preceding expression, we obtain the following solution sets that satisfied the inequality (3.10)

$$\beta_2 > \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 \text{ and } \beta_2 < \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} - \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2,$$

thus, the solution set that adequately satisfies the condition of the problems is

$$\beta_2 > \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2, \tag{3.12}$$

thus obtaining the lower bound for the parameter β_2 in which, f_D in the interval $(\zeta_1, \zeta_2] \subset [0, 1] \subset \mathbb{R}$ has fixed points. Finally, if we combine the inequalities (3.6) and (3.12) we obtain the interval that we desired showing.

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.13)

Proposition 1 defined two fixed points, denoted by $x_{2,1}^*$ and $x_{2,2}^*$ in the interval $\zeta_1 \leq x \leq \zeta_2$ when β_2 fulfills (3.13) as follows

$$x_{2,1}^* = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) - 1 + \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}}{2\gamma \beta_2}, \qquad (3.14)$$

and

$$x_{2,2}^* = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) - 1 - \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}}{2\gamma \beta_2}.$$
 (3.15)

Proposition 2 The fixed point $x_{2,2}^*$ of the bimodal map f_D (3.1) in the interval $\zeta_1 \leq x \leq \zeta_2$ given by (3.15) is always repulsive if

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.16)

Proof. For this proof, it is enough verifying that the following condition $|Df_{D_2}(x_{2,2}^*, \gamma, \beta_2)| > 1$ is fulfilled. If we find the condition for β_2 such that the absolute value of the derivative of f_{D_2} evaluated at the fixed point $x_{2,2}^*$ is greater than one then this fixed point is always repulsive. Thus, we proceed to find the derivative of the (3.3) we obtain

$$Df_{D_2}(x,\gamma,\beta_2) = -\gamma\beta_2[2x - (\zeta_1 + \zeta_2)], \qquad (3.17)$$

by substituting the fixed point (3.15) into (3.17) and applied the condition $|Df_{D_2}(x_{2,2}^*, \gamma, \beta_2)| > 1$ is obtained

$$\left|-\gamma\beta_{2}\left[2x_{2,2}^{*}-(\zeta_{1}+\zeta_{2})\right]\right|>1,$$
 (3.18)

by simplifying the last expression, results

$$\left|\gamma\beta_2\left[\frac{\gamma\beta_2(\zeta_1+\zeta_2)-1-\sqrt{\Delta}-\gamma\beta_2(\zeta_1+\zeta_2)}{\gamma\beta_2}\right]\right|>1,$$

where

$$\Delta = [\gamma \beta_2 (\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2,$$

by simplifying, we have

$$\left|1 + \sqrt{\Delta}\right| = \left|1 + \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}\right| > 1,$$

by properties of absolute value, the preceding expression can expressed into two inequalities that which can be described by (3.19) and (3.20)

$$1 + \sqrt{[\gamma\beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2\beta_2^2\zeta_1\zeta_2} < -1, \qquad (3.19)$$

$$1 + \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2} > 1, \qquad (3.20)$$

in the inequality (3.19) we can check that if we substrates one to both sides, is obtained

$$\sqrt{[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2-4\gamma^2\beta_2^2\zeta_1\zeta_2} < -2,$$

the preceding inequality has not solution in the real numbers field, however, if doing the same processing in the inequality (3.20) we can see that the

solutions of this inequality are included in the real numbers field, thus, we obtain

$$\sqrt{[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2-4\gamma^2\beta_2^2\zeta_1\zeta_2}>0,$$

squaring both sides, results

$$[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2-4\gamma^2\beta_2^2\zeta_1\zeta_2>0,$$

by expanded and group terms, using the parameter β_2 as variable, the inequality (3.10) are obtained. Thus, by solving it using the general formula, we concluded that the solution set for β_2 that satisfying the conditions in our problem was the next

$$\beta_2 > \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2, \tag{3.21}$$

by the last result, we concluded that the fixed point $x_{2,2}^*$ is always repulsive if β_2 satisfies (3.21).

For prove the upper bound of the interval (3.13), we know by the proposition one that β_2 has upper bound described by

$$\beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2},$$

thus, it is enough verifying that take the upper bound of the parameter β_2 , the fixed point $x_{2,2}^*$ of the $\zeta_1 \leq x \leq \zeta_2$ substituted into the f_{D_2} expressed in (3.3) is repulsive. Thus, took the upper bound

$$\beta_{2_M} = \frac{4}{\gamma(\zeta_2 - \zeta_1)^2} \tag{3.22}$$

by substituting the preceding equation into (3.3), we obtained

$$f_{D_2}(x,\gamma,\beta_{2_M}) = \gamma \beta_{2_M}(x-\zeta_1)(\zeta_2-x), \qquad (3.23)$$

when differentiating the preceding equation, results

$$Df_{D_2}(x,\gamma,\beta_{2_M}) = -\gamma\beta_{2_M}[2x - (\zeta_1 + \zeta_2)].$$
(3.24)

Now, by substituting (3.15) into of the derivative (3.24) we obtained

$$Df_{D_2}(x_{2,2}^*,\gamma,\beta_{2_M}) = -2\gamma\beta_{2_M}\Lambda,$$

where

$$\Lambda = \frac{\gamma \beta_{2_M}(\zeta_1 + \zeta_2) - 1 - \sqrt{[\gamma \beta_{2_M}(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_{2_M}^2 \zeta_1 \zeta_2}}{2\gamma \beta_{2_M}},$$

by simplifying the previous expression

$$Df_{D_2}(x_{2,2}^*) = 1 + \sqrt{\left[\frac{4}{(\zeta_2 - \zeta_1)^2}(\zeta_1 + \zeta_2) - 1\right]^2 - \frac{64}{(\zeta_2 - \zeta_1)^4}\zeta_1\zeta_2}.$$
 (3.25)

Now, let us analyze the discriminant to verifying that $\Delta > 0$, we know that

$$\Delta = \left[\frac{4}{(\zeta_2 - \zeta_1)^2}(\zeta_1 + \zeta_2) - 1\right]^2 - \frac{64}{(\zeta_2 - \zeta_1)^4}\zeta_1\zeta_2,$$

by expanding

$$\Delta = \frac{16}{(\zeta_2 - \zeta_1)^4} (\zeta_1 + \zeta_2)^2 - \frac{8}{(\zeta_2 - \zeta_1)^2} (\zeta_1 + \zeta_2) + 1 - \frac{64}{(\zeta_2 - \zeta_1)^2} \zeta_1 \zeta_2,$$

by grouping terms, we obtain

$$\Delta = \frac{16}{(\zeta_2 - \zeta_1)^4} \left[(\zeta_1 + \zeta_2)^2 - 4\zeta_1 \zeta_2 \right] - \frac{8}{(\zeta_2 - \zeta_1)^2} (\zeta_1 + \zeta_2) + 1,$$
$$\Delta = \frac{16}{(\zeta_2 - \zeta_1)^4} (\zeta_2 - \zeta_1)^2 - \frac{8}{(\zeta_2 - \zeta_1)^2} (\zeta_1 + \zeta_2) + 1,$$

by simplifying

$$\Delta = \frac{16}{(\zeta_2 - \zeta_1)^2} - \frac{8}{(\zeta_2 - \zeta_1)^2}(\zeta_1 + \zeta_2) + 1.$$

Finally, we obtain that

$$\Delta = \frac{8}{(\zeta_2 - \zeta_1)^2} \left[2 - (\zeta_1 + \zeta_2) \right] + 1,$$

and since $\zeta_1 < \zeta_2$ and $\zeta_2 \le 1$ then $\zeta_1 + \zeta_2 < 2$ therefore

$$\Delta = \frac{8}{(\zeta_2 - \zeta_1)^2} \left[2 - (\zeta_1 + \zeta_2) \right] + 1 > 0.$$

Based on the previous analysis, we conclude that

$$\Delta = \left[\frac{4}{(\zeta_2 - \zeta_1)^2}(\zeta_1 + \zeta_2) - 1\right]^2 - \frac{64}{(\zeta_2 - \zeta_1)^4}\zeta_1\zeta_2 > 0,$$

and therefore, (3.25) is always upper than one. Consequently, if the parameter β_2 took the upper bound expressed by

$$\beta_2 = \frac{4}{\gamma(\zeta_2 - \zeta_1)^2},$$

then, the fixed point

$$x_{2,2}^* = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) - 1 - \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}}{2\gamma \beta_2},$$

satisfies that $Df_{D_2}(x_{2,2}^*) > 1$. Therefore, the fixed point is repulsive if (3.16) is satisfied.

Proposition 3 Consider the bimodal map given by (3.1). The fixed point $x_{2,1}^*$ in the interval $\zeta_1 \leq x \leq \zeta_2$ expressed by (3.14) is attractive if:

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 < \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}, \tag{3.26}$$

and repulsive if:

$$\frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.27)

Proof. For find the interval where the parameter β_2 makes that the fixed point $x_{2,1}^*$ is repulsive in the interval $[\zeta_1, \zeta_2]$ is repulsive, we use the condition that $|Df_{D_2}(x_{2,1}^*, \gamma, \beta_2)| < 1$, thus, the derivative of (3.3) results

$$Df_{D_2}(x, \gamma, \beta_2) = -\gamma \beta_2 [2x - (\zeta_1 + \zeta_2)].$$

By substituting the fixed point $x_{2,1}^*$ shown in (3.14) into the preceding expression, by simplifying we obtain

$$Df_{D_2}(x_{2,1}^*,\gamma,\beta_2) = \left| 1 - \sqrt{[\gamma\beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2\beta_2^2\zeta_1\zeta_2} \right|.$$

Now, we choose a $\varepsilon > 0$ and the lower bound of the interval for the parameter β_2 that which is expressed by (3.26). The lower bound will be denoted by

$$\beta_m = \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2. \tag{3.28}$$

An analogously way to write the parameter β_m can be seen below

$$\beta_m = \frac{\zeta_1 + \zeta_2 + 2\sqrt{\zeta_1 \zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.29)

Previously, we mention an ε , this is mentioned with the purpose that the parameter β_2 can not took the upper and lower bounded, since the interval shown in (3.26) is open and consequently, this interval not include the boundaries. Now, the purpose is found the lower and upper value that ε can take it, for this, we know, by the interval expressed in (3.26) the upper bound of β_2 , that which will be denoted by β_M and is expressed by

$$\beta_M = \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.30)

Thus, we can write (3.30) of the following way

$$\beta_M = \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 - \zeta_2)^2 + \zeta_1 \zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Now, let β_0 the parameter shown in (3.29) but adding ε within of the radical, results

$$\beta_0 = \frac{\zeta_1 + \zeta_2 + 2\sqrt{\zeta_1\zeta_2 + \varepsilon}}{\gamma(\zeta_2 - \zeta_1)^2} \tag{3.31}$$

such that if we want to find the interval where ε can be chosen arbitrarily, it is enough to determine the interval of values that ε must take on β_0 in order to satisfy

$$\beta_m < \beta_0 < \beta_M. \tag{3.32}$$

By substituting the previously denoted values, we obtain

$$\frac{\zeta_1 + \zeta_2 + 2\sqrt{\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \frac{\zeta_1 + \zeta_2 + 2\sqrt{\zeta_1\zeta_2 + \varepsilon}}{\gamma(\zeta_2 - \zeta_1)^2} < \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 - \zeta_2)^2 + \zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.33)

By simplifying the preceding inequality, it results

$$\sqrt{\zeta_1\zeta_2} < \sqrt{\zeta_1\zeta_2 + \varepsilon} < \sqrt{(\zeta_1 - \zeta_2)^2 + \zeta_1\zeta_2},$$

analogously, we obtain

$$\zeta_1\zeta_2 < \zeta_1\zeta_2 + \varepsilon < (\zeta_1 - \zeta_2)^2 + \zeta_1\zeta_2.$$

Finally, results

$$0 < \varepsilon < (\zeta_1 - \zeta_2)^2.$$
 (3.34)

The preceding interval is the interval where ε can be chosen arbitrarily. now, we proceed to verify if the following condition $|Df_{D_2}(x_{2,1}^*, \gamma, \beta_2)| < 1$ is satisfying. Previously was obtain the following expression

$$Df_{D_2}(x_{2,1}^*,\gamma,\beta_2) = \left| 1 - \sqrt{\gamma^2(\zeta_2 - \zeta_1)^2 \beta_2^2 - 2\gamma(\zeta_1 + \zeta_2)\beta_2 + 1} \right|.$$
(3.35)

The preceding expression is the result of applied the condition

$$\left| Df_{D_2}(x_{2,1}^*, \gamma, \beta_2) \right| < 1.$$

By simplifying and substituting β_0 into β_2 , it is obtained

$$Df_{D_2}(x_{2,1}^*,\gamma,\beta_0) = \left|1-\sqrt{\Theta}\right|,$$

where

$$\Theta = \gamma^2 (\zeta_2 - \zeta_1)^2 \beta_0^2 - 2\gamma (\zeta_1 + \zeta_2) \beta_0 + 1,$$

and β_0 is defined in (3.31). By substituting β_0 in Θ and simplifying we have

$$\Theta = \frac{(\zeta_2 - \zeta_1)^2 + 4\zeta_1\zeta_2 + 4\varepsilon - (\zeta_1 + \zeta_2)^2}{(\zeta_2 - \zeta_1)^2} = \frac{4\varepsilon}{(\zeta_2 - \zeta_1)^2}$$

Thus, the expression $Df_{D_2}(x_{2,1}^*, \gamma, \beta_0) = \left|1 - \sqrt{\Theta}\right|$ results

$$Df_{D_2}(x_{2,1}^*, \gamma, \beta_0) = \left| 1 - \sqrt{\frac{4\varepsilon}{(\zeta_2 - \zeta_1)^2}} \right|.$$

Finally, it results

$$Df_{D_2}(x_{2,1}^*, \gamma, \beta_0) = \left|1 - \frac{2\sqrt{\varepsilon}}{(\zeta_2 - \zeta_1)}\right| < 1,$$

from the last expression, we can see that for each ε taken from the interval $0 < \varepsilon < (\zeta_1 - \zeta_2)^2$ it holds that $|Df_{D_2}(x_{2,1}^*, \gamma, \beta_2)| < 1$ and how the preceding interval was found using the upper and lower bounds of the parameter

 β_2 expressed in (3.26), this shows that for each value of the parameter β_2 contained in

$$\frac{(\sqrt{\zeta_1} + \sqrt{\zeta_2})^2}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 < \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.36)

The fixed point $x_{1,2}^*$ is always attractive. Now, to demonstrate the interval for the parameter β_2 where the fixed point $x_{2,1}^*$, which is included in the interval $\zeta_1 \leq x \leq \zeta_2$, is always repulsive, we use the following condition:

$$\left| Df_{D_2}(x_{2,1}^*, \gamma, \beta_2) \right| > 1.$$
 (3.37)

Thus, by substituting the fixed point denoted by (3.14) into the derivative of f_{D_2} as expressed by (3.17) and applying the condition (3.37), it results

$$\left|1 - \sqrt{[\gamma \beta_2(\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2}\right| > 1.$$
(3.38)

The preceding inequality is fulfilled if

$$\sqrt{[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2 - 4\gamma^2\beta_2^2\zeta_1\zeta_2} > 2.$$
(3.39)

By squaring both sides of the previous inequality, is obtained

$$[\gamma\beta_2(\zeta_1+\zeta_2)-1]^2-4\gamma^2\beta_2^2\zeta_1\zeta_2>4.$$

By expanding, results

$$\gamma^2 (\zeta_2 - \zeta_1)^2 \beta_2^2 - 2\gamma (\zeta_1 + \zeta_2) \beta_2 + 1 > 4.$$
(3.40)

By subtracting four from both sides, it results in

$$\gamma^2(\zeta_2 - \zeta_1)^2 \beta_2^2 - 2\gamma(\zeta_1 + \zeta_2)\beta_2 - 3 > 0.$$
(3.41)

By solving the previous inequality using the general, produces

$$\beta_2 = \frac{\gamma(\zeta_1 + \zeta_2) \pm \sqrt{\gamma^2(\zeta_1 + \zeta_2)^2 - (-3)(\gamma^2(\zeta_2 - \zeta_1)^2)}}{\gamma^2(\zeta_2 - \zeta_1)^2}.$$
 (3.42)

By simplifying the previous expression

$$\beta_2 = \frac{(\zeta_1 + \zeta_2) \pm \sqrt{(\zeta_1 + \zeta_2)^2 + 3(\zeta_2 - \zeta_1)^2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Analogously, we have

$$\beta_2 = \frac{(\zeta_1 + \zeta_2) \pm 2\sqrt{\zeta_1^2 - \zeta_1 \zeta_2 + \zeta_2^2}}{\gamma(\zeta_2 - \zeta_1)^2},$$

when factoring the term inside the square root, it results in

$$\beta_2 = \frac{(\zeta_1 + \zeta_2) \pm 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Thus, the solution set are given by

$$\beta_2 > \frac{(\zeta_1 + \zeta_2) + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2},$$
(3.43)

$$\beta_2 < \frac{(\zeta_1 + \zeta_2) - 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.44)

However, the solution set that satisfies our conditions is

$$\beta_2 > \frac{(\zeta_1 + \zeta_2) + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}, \tag{3.45}$$

that which is the lower bound of the interval shown in (3.27). Now, to demonstrate the upper bound it is enough notice the inequality (3.41) and we defined by \mathcal{G} the left term, thus

$$\mathcal{G}(\beta_2) = \gamma^2 (\zeta_2 - \zeta_1)^2 \beta_2^2 - 2\gamma (\zeta_1 + \zeta_2) \beta_2 - 3.$$

Notice that $\mathcal{G}(\beta_2)$ is a continuous function, therefore, it was previously shown that when β_2 takes the lower bound, it makes that $\mathcal{G}(\beta_2) > 1$ and how the upper bound of the parameter β_2 shown in the interval defined by (3.27) satisfies that

$$\beta_2 = \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.46)

Therefore, due to continuity of $\mathcal{G}(\beta_2) > 1$, when β_2 reaches the upper bound, it ensures that $\mathcal{G}(\beta_2) > 1$, and it can be concluded that the fixed point $x_{2,1}^*$ included in $\zeta_1 \leq x \leq \zeta_2$ is always repulsive if β_2 takes values within the following interval

$$\frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}$$
(3.47)

Proposition 4 Consider the bimodal map (3.1). A saddle-node bifurcation occurs in the subdomain $\zeta_1 \leq x \leq \zeta_2$ when

$$\beta_0 = \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2. \tag{3.48}$$

Proof. To demonstrate this Proposition, we need to show that every point stated in Definition 36 holds true. Regarding the first point of the aforementioned definition, it is necessary to demonstrate that prior to β_0 , the bimodal map f_D , specifically the unimodal map f_{D_2} , does not possess fixed points. To prove this, let's observe that the discriminant Δ of the fixed points $x_{2,1}^*$ and $x_{2,2}^*$, defined by (3.14) and (3.15) respectively, where:

$$\Delta = [\gamma \beta_2 (\zeta_1 + \zeta_2) - 1]^2 - 4\gamma^2 \beta_2^2 \zeta_1 \zeta_2.$$

Notice, from Proposition 1, that discriminant satisfies $\Delta > 0$ for any

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Moreover, notice that $\Delta = 0$ when $\beta_2 = \beta_0$. Therefore, for any $\beta_2 = \beta_0 - \varepsilon$ with $\varepsilon > 0$, it satisfies that $\Delta < 0$, therefore, f_{D_1} has not fixed points, fulfilling issue 1 of Definition 36.

For demonstrates the issue 2 of Definition 36, it is enough to notice that when $\beta_2 = \beta_0$ the discriminant $\Delta = 0$, which implies that the fixed points $x_{2,1}^*$ and $x_{2,2}^*$ satisfy $x_{2,2}^* = x_{2,1}^*$. Therefore, f_{D_2} has only one fixed point, fulfilling issue 2 of Definition 36.

Finally, to demonstrate issue 3 of the Definition 36, it is sufficient to consider Propositions 2 and 3, to state when β_2 satisfies the condition

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 < \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2},$$

there exist two fixed points $x_{2,1}^*$ and $x_{2,2}^*$, where the $x_{2,2}^*$ is a repulsive fixed point and $x_{2,1}^*$ is an invariant point.

Proposition 5 Consider the bimodal map (3.1). A period-doubling bifurcation occurs in the interval $\zeta_1 \leq x \leq \zeta_2$ when:

$$\beta_0 = \frac{(\zeta_1 + \zeta_2) + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2}.$$
(3.49)

Proof. In the proof of proposition 3, we demonstrated issues 1 and 2 of Definition 37. When it was concluded that for some $\beta_2 < \beta_0$, the fixed point $x_{2,1}^*$ in the interval $\zeta_1 \leq x \leq \zeta_2$, defined in (3.14), was attractive, and for some $\beta_2 > \beta_0$, this fixed point was repulsive. Now, we must demonstrate issues 3 and 4. For point 3, we need to show the existence of a 2-cycle. To demonstrate this, we use

$$f_{D_2}(f_{D_2}(x,\gamma,\beta_2)) = f_{D_2}^2(x,\gamma,\beta_2) = x.$$
(3.50)

We know that

$$f_{D_2}(x,\gamma,\beta_2) = -\gamma\beta_2[x^2 - (\zeta_1 + \zeta_2)x + \zeta_1\zeta_2], \qquad (3.51)$$

of the previous equation, we define

$$\gamma\beta_2 = r$$
, $\zeta_1 + \zeta_2 = a$ and $\zeta_1\zeta_2 = b$.

By substituting the previous defined parameters into (3.51)

$$f_{D_2}(x, r, a, b) = -rx^2 + rax - rb.$$
(3.52)

Thus, if we applied $f_{D_2}(f_{D_2}(x,r,a,b)) = f_{D_2}^2(x,r,a,b) = x$ it results

$$f_{D_2}^2(x,r,a,b) = -r(-rx^2 + rax - rb)^2 + ra(-rx^2 + rax - rb) - rb = x.$$
(3.53)

By expanding and simplifying

$$f_{D_2}^2(x, r, a, b) - x = -r^3 x^4 + 2r^3 a x^3 - r^2 [r(2b+a^2)+a)]x^2 + (2r^3 a b + r^2 a^2 - 1)x - rb(1+ra+r^2b).$$

From the previous equation, we know that the fixed points denoted by (3.14) and (3.15) are solution of $f_{D_2}^2(x, r, a, b) - x=0$, and we also know that these fixed points are determined by $f_{D_2}(x, r, a, b) - x$, so, if we make

$$\frac{f_{D_2}^2(x,r,a,b) - x}{f_{D_2}(x,r,a,b) - x} = -\frac{f_{D_2}^2(x,r,a,b) - x}{rx^2 + (1 - ra)x + rb},$$
(3.54)

by dividing, it results

$$-\frac{f_{D_2}^2(x,r,a,b)-x}{rx^2+(1-ra)x+rb} = rx^2 - r(ra+1)x + (r^2b+ra+1).$$
(3.55)

By setting to zero and returning to the previous parameters, we have

$$\gamma \beta_2 x^2 - \gamma \beta_2 [\gamma \beta_2 (\zeta_1 + \zeta_2) + 1] x + [\gamma^2 \beta_2^2 \zeta_1 \zeta_2 + \gamma \beta_2 (\zeta_1 + \zeta_2) + 1] = 0. \quad (3.56)$$

Thus, the solutions of the preceding equation are the periodic points with period 2 of $f_{D_2}(x, \gamma, \beta_2)$. By solving using the general formula, we have

$$x_1 = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) + 1 + \sqrt{\gamma^2 \beta_2^2(\zeta_2 - \zeta_1)^2 - 2\gamma \beta_2(\zeta_1 + \zeta_2) - 3}}{2\gamma \beta_2}, \qquad (3.57)$$

$$x_2 = \frac{\gamma \beta_2(\zeta_1 + \zeta_2) + 1 - \sqrt{\gamma^2 \beta_2^2(\zeta_2 - \zeta_1)^2 - 2\gamma \beta_2(\zeta_1 + \zeta_2) - 3}}{2\gamma \beta_2}.$$
 (3.58)

We can notice that the expressions (3.57) and (3.58) depend on β_2 ; therefore, if the discriminant Δ satisfies that $\Delta > 0$, then x_1 and x_2 belong to \mathbb{R} , where

$$\Delta = \gamma^2 \beta_2^2 (\zeta_2 - \zeta_1)^2 - 2\gamma \beta_2 (\zeta_1 + \zeta_2) - 3.$$
(3.59)

Thus, we have

$$\gamma^2 (\zeta_2 - \zeta_1)^2 \beta_2^2 - 2\gamma (\zeta_1 + \zeta_2) \beta_2 - 3 > 0.$$
(3.60)

By solving the preceding inequality using the general formula, we have

$$\beta_2 = \frac{\gamma(\zeta_1 + \zeta_2) \pm \sqrt{\gamma^2(\zeta_1 + \zeta_2)^2 + 3\gamma^2(\zeta_2 - \zeta_1)^2}}{\gamma^2(\zeta_2 - \zeta_1)^2}.$$

By simplifying, we have

$$\beta_2 = \frac{\zeta_1 + \zeta_2 \pm \sqrt{(\zeta_1 + \zeta_2)^2 + 3(\zeta_2 - \zeta_1)^2}}{\gamma(\zeta_2 - \zeta_1)^2},$$
$$\beta_2 = \frac{\zeta_1 + \zeta_2 \pm 2\sqrt{\zeta_1^2 - \zeta_1\zeta_2 + \zeta_2^2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Finally, it results

$$\beta_2 = \frac{\zeta_1 + \zeta_2 \pm 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Therefore, the solution sets are

$$\beta_2 > \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}, \quad \beta_2 < \frac{\zeta_1 + \zeta_2 - 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$

However, the solution set that satisfies the conditions of β_2 is

$$\beta_2 > \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.61)

Consequently, if β_2 satisfies the preceding condition then there is a 2-cycle and by the proof of the proposition 3, the fixed point $x_{2,1}^*$ is a repulsive fixed point when

$$\beta_2 > \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.62)

Now, we must demonstrate that

$$|Df_{D_2}(x_1)Df_{D_2}(x_2)| < 1.$$

By substituting (3.57) and (3.58) into (3.17), and by simplifying we have

$$\left| (1 + \sqrt{\Delta})(1 - \sqrt{\Delta}) \right|, \tag{3.63}$$

where

$$\Delta = \gamma^2 \beta_2^2 (\zeta_2 - \zeta_1)^2 - 2\gamma \beta_2 (\zeta_1 + \zeta_2) - 3.$$

Now, by expanding (3.63)

 $|1-\Delta|.$

We notice that $\Delta > 0$ when β_2 satisfies (3.47), which includes the condition (3.62), then, we concluded that

$$|1 - \Delta| < 1.$$

However, by absolute value properties, we have

$$-1 < 1 - \Delta < 1,$$

analogously

 $0 < \Delta < 2,$

from $\Delta > 0$ it results

$$\beta_2 > \frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.64)

And from $\Delta < 2$ we have

$$0 < \beta_2 < \frac{\zeta_1 + \zeta_2 + \sqrt{6(\zeta_1 + \zeta_2)^2 - 20\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
(3.65)

Consequently, from (3.64) and (3.65) results

$$\frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 < \frac{\zeta_1 + \zeta_2 + \sqrt{6(\zeta_1 + \zeta_2)^2 - 20\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.66)

Since $\zeta_1 < \zeta_2$ and $\zeta_2 \leq 1$, we know that $\zeta_1 + \zeta_2 < 2$ and $\zeta_1 \zeta_2 < 1$ then, from the preceding we have

$$\beta_2 < \frac{\zeta_1 + \zeta_2 + \sqrt{6(\zeta_1 + \zeta_2)^2 - 20\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Therefore, we can notice that the following condition $|Df_{D_2}(x_1)Df_{D_2}(x_2)| < 1$ is satisfied if the parameter β_2 takes values within of

$$\frac{\zeta_1 + \zeta_2 + 2\sqrt{(\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 < \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.67)

Finally, issue 3 of Definition 37 has been demonstrated. Now, to demonstrate issue 4 of 37, it is enough to show that

$$\lim_{\beta_2 \to \beta_0} x_i = x_{2,1}^*, \tag{3.68}$$

that is

$$\lim_{\beta_2 \to \beta_0} \frac{\gamma \beta_2(\zeta_1 + \zeta_2) + 1 \pm \sqrt{\gamma^2 \beta_2^2 (\zeta_2 - \zeta_1)^2 - \gamma \beta_2 (\zeta_1 + \zeta_2) - 3}}{2\gamma \beta_2} = x_{2,1,\beta_0}^*, \qquad (3.69)$$

where $x_{2,1,\beta_0}^*$ is the fixed point of the interval $\zeta_1 \leq x \leq \zeta_2$, as shown in (3.14). By substituting β_0 into β_2

$$x_{2,1,\beta_0}^* = \frac{\gamma\beta_0(\zeta_1 + \zeta_2) - 1 + \sqrt{(\gamma\beta_0(\zeta_1 + \zeta_2) - 1)^2 - 4\gamma^2\beta_0^2\zeta_1\zeta_2}}{2\gamma\beta_0}, \quad (3.70)$$

where β_0 is defined in (3.49), we can notice that

$$\lim_{\beta_2 \to \beta_0} \sqrt{\gamma^2 \beta_2^2 (\zeta_2 - \zeta_1)^2 - \gamma \beta_2 (\zeta_1 + \zeta_2) - 3} \to 0.$$

Thus, by simplifying (3.69) we have

$$\lim_{\beta_2 \to \beta_0} \frac{\gamma \beta_2(\zeta_1 + \zeta_2) + 1}{2\gamma \beta_2} = x_{2,1,\beta_0}^*.$$
(3.71)

From (3.70) we have

$$x_{2,1,\beta_0}^* = \frac{\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right) (\zeta_1 + \zeta_2) - 1 + \sqrt{\Pi}}{2\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right)},$$
(3.72)

where

$$\alpha = (\zeta_1 + \zeta_2)^2 - 3\zeta_1\zeta_2, \tag{3.73}$$

and

$$\Pi = \gamma^2 (\zeta_2 - \zeta_1)^2 \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2} \right)^2 - 2\gamma(\zeta_1 + \zeta_2) \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2} \right) + 1.$$
(3.74)

By expanding the previous equation

$$\Pi = \frac{(\zeta_1 + \zeta_2)^2 + 4(\zeta_1 + \zeta_2)\sqrt{\alpha} + 4\alpha - 2(\zeta_1 + \zeta_1)^2 - 4(\zeta_1 + \zeta_2)\sqrt{\alpha} + (\zeta_2 - \zeta_1)^2}{(\zeta_2 - \zeta_1)^2}.$$

Finally, by simplifying

$$\Pi = \frac{-(\zeta_1 + \zeta_2)^2 + 4\alpha + (\zeta_2 - \zeta_1)^2}{(\zeta_2 - \zeta_1)^2} = \frac{3(\zeta_1 + \zeta_2)^2 + (\zeta_2 - \zeta_1)^2 - 12\zeta_1\zeta_2}{(\zeta_2 - \zeta_1)^2} = 4.$$

By substituting Π into (3.72)

$$x_{2,1,\beta_0}^* = \frac{\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right) (\zeta_1 + \zeta_2) - 1 + \sqrt{4}}{2\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right)}.$$
 (3.75)

Finally after simplification, we obtain the following solution

$$x_{2,1,\beta_0}^* = \frac{\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right)(\zeta_1 + \zeta_2) + 1}{2\gamma \left(\frac{\zeta_1 + \zeta_2 + 2\sqrt{\alpha}}{\gamma(\zeta_2 - \zeta_1)^2}\right)} = \frac{\gamma \beta_0(\zeta_1 + \zeta_2) + 1}{2\gamma \beta_0}.$$
 (3.76)

Now, from the expression (3.71) we have

$$\lim_{\beta_2 \to \beta_0} \frac{\gamma \beta_2(\zeta_1 + \zeta_2) + 1}{2\gamma \beta_2} = \frac{\gamma \beta_0(\zeta_1 + \zeta_2) + 1}{2\gamma \beta_0} = x_{2,1,\beta_0}^*.$$
 (3.77)

Therefore, it is demonstrated that when $\beta_2 \to \beta_0$ then $x_i \to x_{2,1,\beta_0}^*$ for i = 1, 2, and based on the above, we conclude the proof of the four issues of Definition 37, and we can assert the existence of a period-doubling bifurcation.

Proposition 6 The difference map shown in (3.1) exhibits a trapping region or an invariant set $\mathcal{A}_2 \subset (\zeta_1, \zeta_2] \subset (0, 1]$ if

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 \le \beta_2 \le \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}, \qquad (3.78)$$

where

$$\mathcal{A}_{2} = \left[x_{2,2}^{*}, \frac{\gamma \beta_{2}(\zeta_{1} + \zeta_{2}) + 1 + \sqrt{(\gamma \beta_{2} \zeta_{1} + \zeta_{2}) - 1)^{2} - 4\gamma^{2} \beta_{2}^{2} \zeta_{1} \zeta_{2}}}{\gamma \beta_{2}} \right], \quad (3.79)$$

and $x_{2,2}^*$ denotes the fixed point of the interval $\zeta_1 \leq x \leq \zeta_2$ as defined by (3.15).

Proof. To demonstrate this proposition, it is enough to find the upper value of the parameter β_2 such that the second iteration of f_{D_2} with x_{m_2} as the initial condition x_0 is higher than the fixed point $x_{2,2}^*$, where x_{m_2} denotes the midpoint of the interval $\zeta_1 \leq x \leq \zeta_2$ and is defined by

$$x_{m_2} = \frac{\zeta_1 + \zeta_2}{2}.$$
 (3.80)

The previous explanation can be resumed by finding a certain value for β_2 such that

$$f_{D_2}^2(x_{m_2}, \gamma, \beta_2) \ge x_{2,2}^*.$$
 (3.81)

This proposition will be demonstrated; thus, we have

$$f_{D_2}(x,\gamma,\beta_2) = -\gamma\beta_2[x^2 - (\zeta_1 + \zeta_2)x + \zeta_1\zeta_2].$$
 (3.82)

We use

$$\gamma \beta_2 = r, \quad \zeta_1 + \zeta_2 = a \text{ and } \zeta_1 \zeta_2 = b.$$
 (3.83)

By expanding (3.81) and by substituting the terms defined in (3.83), it results

$$f_{D_2}(x, r, a, b) = -rx^2 + rax - rb.$$
 (3.84)

By substituting x_{m_2} into (3.84) and by simplifying, we have

$$f_{D_2}(x_{m_2}, r, a, b) = r \frac{(a^2 - 4b)}{4}.$$
 (3.85)

When performing the second iteration $f_{D_2}^2(x_{m_2}, r, a, b)$ we have

$$f_{D_2}^2(x_{m_2}, r, a, b) = -r\left(r\frac{(a^2 - 4b)}{4}\right)^2 + ra\left(r\frac{(a^2 - 4b)}{4}\right) - rb.$$
(3.86)

If we defined

$$c = a^2 - 4b, (3.87)$$

then, by expanding (3.86) and by substituting (3.87), it results

$$f_{D_2}^2(x_{m_2}, r, a, b) = \frac{-r^3 c^2 + 4r^2 a c - 16rb}{16}.$$
 (3.88)

Now, we know that the fixed point $x_{2,2}^*$ as defined in (3.15) is

$$x_{2,2}^* = \frac{\gamma\beta_2(\zeta_1 + \zeta_2) - 1 - \sqrt{(\gamma\beta_2(\zeta_1 + \zeta_2) - 1)^2 - 4\gamma^2\beta_2^2\zeta_1\zeta_2}}{2\gamma\beta_2}$$

The previous expression by substituting the parameters defined in (3.83) becomes

$$x_{2,2}^* = \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}.$$
(3.89)

Applying the condition shown in (3.81), it results

$$\frac{-r^3c^2 + 4r^2ac - 16rb}{16} \ge \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}.$$
 (3.90)

By simplifying

$$-r^{4}c^{2} + 4r^{3}ac - 16r^{2}b \ge 8(ra - 1 - \sqrt{(ra - 1)^{2} - 4r^{2}b})$$

Finally, we have

$$r^{4}c^{2} - 4r^{3}ac + 16r^{2}b + 8ra - 8 \le 8\sqrt{(ra-1)^{2} - 4r^{2}b}.$$

By squaring both sides, it results

$$(r^4c^2 - 4r^3ac + 16r^2b + 8ra - 8)^2 \le (8\sqrt{(ra-1)^2 - 4r^2b})^2.$$

By expanding and simplifying both sides, we have

$$\begin{aligned} c^4r^8 - 8ac^3r^7 + 16c^2(2b+a^2)r^6 + 16ac(c-8b)r^5 - \\ & 16(c^2 + 4ac^2 - 16b^2)r^4 + 64a(c+4b)r^3 \leq 0. \end{aligned}$$

By dividing the previous expression by r^3 , becomes

$$c^{4}r^{5} - 8ac^{3}r^{4} + 16c^{2}(2b+a^{2})r^{3} + 16ac(c-8b)r^{2} - 16(c^{2} + 4ac^{2} - 16b^{2})r + 64a(c+4b) \le 0, \quad (3.91)$$

when factoring the previous expression, it results in

$$(cr^{2} - 2ar - 8)(c^{3}r^{3} - 6ac^{2}r^{2} + 4c[8b + a^{2} + 2c]r - 8a[2c + 8b - a^{2}]) \le 0.$$
(3.92)

In the previous expression, we take r as the variable, which is the term containing β_2 . The third-degree factor, when solved in terms of r, yields two complex roots and one real root; however, none of them satisfy the requirements for this study. Now, for the second-degree factor, we obtain the inequality

$$(cr^2 - 2ar - 8) \le 0. \tag{3.93}$$

By solving using the general formula

$$r = \frac{a \pm \sqrt{a^2 + 8c}}{c}.\tag{3.94}$$

By substituting $c = (a^2 - 4b)$ we have

$$r = \frac{a \pm \sqrt{a^2 + 8(a^2 - 4b)}}{a^2 - 4b}$$

By simplifying

$$r = \frac{a \pm \sqrt{9a^2 - 32b}}{a^2 - 4b}$$

We know that $a = \zeta_1 + \zeta_2$, $b = \zeta_1 \zeta_2$ and $r = \gamma \beta_2$, thus, the preceding expression becomes

$$\gamma \beta_2 = \frac{(\zeta_1 + \zeta_2) \pm \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{(\zeta_1 + \zeta_2)^2 - 4\zeta_1\zeta_2}$$

Solving for β_2 and simplifying

$$\beta_2 = \frac{(\zeta_1 + \zeta_2) \pm \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2}$$

Thus, we obtained the interval for β_2 that satisfies $f_{D_2}^2(x_{m_2}, \gamma, \beta_2) \ge x_{2,2}^*$; this interval is

$$\frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2} \le \beta_2 \le \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2}.$$
 (3.95)

However, we can notice from (3.94) that the lower bound of the interval (3.95)

$$\frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2} < 0.$$

Furthermore, from Proposition 1, we know the minimum bound of the parameter β_2 for which fixed points exist in the interval $\zeta_1 \leq x \leq \zeta_2$, and this minimum bound satisfies

$$\frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_1 - \zeta_2)^2} < \frac{(\sqrt{\zeta_1} + \sqrt{\zeta_2})^2}{\gamma(\zeta_2 - \zeta_1)^2}.$$

Thus, by the continuity of f_{D_2} , it is shown that the interval for β_2 will be defined by

$$\frac{(\sqrt{\zeta_1} + \sqrt{\zeta_2})^2}{\gamma(\zeta_2 - \zeta_1)^2} \le \beta_2 \le \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.96)

So, the interval of β_2 defined in (3.96) makes the set $\mathcal{A}_2 \subset (\zeta_1, \zeta_2] \subset (0, 1]$ previously defined as

$$\mathcal{A}_{2} = \left[x_{2,2}^{*}, \frac{\gamma \beta_{2}(\zeta_{1} + \zeta_{2}) + 1 + \sqrt{(\gamma \beta_{2} \zeta_{1} + \zeta_{2}) - 1)^{2} - 4\gamma^{2} \beta_{2}^{2} \zeta_{1} \zeta_{2}}}{\gamma \beta_{2}} \right], \quad (3.97)$$

let be a trapping region or an invariant set.

Proposition 7 The difference map shown in (3.1) exhibits a trapping region or an invariant set $A_1 = [\zeta_0, \zeta_1) = [0, 1/2)$ if

$$\frac{1}{\gamma\zeta_1} \le \beta_1 \le \frac{4}{\gamma\zeta_1}.\tag{3.98}$$

Proof. To demonstrate this proposition, it is enough to find the upper value of the parameter β_1 such that the second iteration of f_{D_1} with x_{m_1} as the initial condition x_0 is higher than the lower bound of \mathcal{A}_1 . That is $\zeta_0 = 0$, where x_{m_1} denotes the midpoint of the interval $\zeta_0 \leq x < \zeta_1$ and is defined by

$$x_{m_1} = \frac{\zeta_0 + \zeta_1}{2} = \frac{\zeta_1}{2}.$$
(3.99)

The previous explanation can be resumed by finding a certain value for β_1 such that

$$f_{D_1}^2(x_{m_1}, \gamma, \beta_1) \ge 0, \tag{3.100}$$

and this proposition will be demonstrated; thus, we have

$$f_{D_1}(x,\gamma,\beta_1) = -\gamma\beta_1(x^2 - \zeta_1 x).$$
 (3.101)

By expanding (3.101) it results

$$f_{D_1}(x,\gamma,\beta_1) = -\gamma\beta_1 x^2 + \gamma\beta_1\zeta_1 x. \qquad (3.102)$$

By substituting x_{m_1} into (3.103) and by simplifying, we have

$$f_{D_1}(x_{m_1}, \gamma, \beta_1) = \frac{\gamma \beta_1 \zeta_1^2}{4}.$$
 (3.103)

When performing the second iteration $f_{D_2}^2(x_{m_2}, \gamma, \beta_1)$ we have

$$f_{D_1}^2(x_{m_1},\gamma,\beta_1) = -\gamma\beta_1 \left(\frac{\gamma\beta_1\zeta_1^2}{4}\right)^2 + \gamma\beta_1\zeta_1 \left(\frac{\gamma\beta_1\zeta_1^2}{4}\right). \tag{3.104}$$

Then, by expanding (3.104), it results

$$f_{D_2}^2(x_{m_2},\gamma,\beta_1) = \frac{-\gamma^3 \beta_1^3 \zeta_1^4 + 4\gamma^2 \beta_1^2 \zeta_1^3}{16}.$$
 (3.105)

By dividing by $\gamma^2 \beta_1^2 \zeta_1^3$ we obtain

$$f_{D_2}^2(x_{m_2},\gamma,\beta_1) = \frac{-\gamma\zeta_1\beta_1 + 4}{16}.$$

Applying the condition shown in (3.100), it result

$$\frac{-\gamma\zeta_1\beta_1+4}{16} \ge \zeta_0.$$

Analogously

$$\frac{-\gamma\zeta_1\beta_1+4}{16} \ge 0.$$

By simplifying

$$-\gamma\zeta_1\beta_1 + 4 \ge 0.$$

Finally, by solving for β_1 we have

$$\beta_1 \le \frac{4}{\gamma \zeta_1}.\tag{3.106}$$

That is the upper bound of the interval of β_1 expressed in (3.98). To demonstrate the lower bound, we proceed to find the fixed points of $f_{D_1}(x, \gamma, \beta_1)$, this is

$$f_{D_1}(x,\gamma,\beta_1) = -\gamma\beta_1 x^2 + \gamma\beta_1 \zeta_1 x = x.$$
 (3.107)

By simplifying we have

$$f_{D_1}(x,\gamma\beta_1) = -\gamma\beta_1 x^2 + (\gamma\beta_1\zeta_1 - 1)x = 0.$$
 (3.108)

By solving using the general formula, we have

$$x_{1,1}^* = \frac{\gamma \beta_1 \zeta_1 - 1 + \sqrt{(\gamma \beta_1 \zeta_1 - 1)^2}}{2\gamma \beta_1}, \quad x_{1,2}^* = \frac{\gamma \beta_1 \zeta_1 - 1 - \sqrt{(\gamma \beta_1 \zeta_1 - 1)^2}}{2\gamma \beta_1}.$$

Finally, we have the fixed points

$$x_{1,1}^* = \frac{\gamma \beta_1 \zeta_1 - 1}{\gamma \beta_1}, \qquad (3.109)$$

$$x_{1,2}^* = 0. (3.110)$$

However, the fixed point must satisfy $x_{1,1}^* \ge 0$, thus

$$\frac{\gamma\beta_1\zeta_1 - 1}{\gamma\beta_1} \ge 0. \tag{3.111}$$

When solving for β_1 , the results is

$$\beta_1 \ge \frac{1}{\gamma \zeta_1}.\tag{3.112}$$

The preceding result is the lower bound of the interval for β_1 defined in (3.98). Therefore, it is shown that \mathcal{A}_1 is a trapping region or an invariant set when β_2 takes values within the interval

$$\frac{1}{\gamma\zeta_1} \le \beta_1 \le \frac{4}{\gamma\zeta_1}.\tag{3.113}$$

Proposition 8 The bimodal map f_D given by (3.1) exhibits monostability in the interval $\zeta_1 \leq x \leq \zeta_2$ if

$$\frac{4x_{2,2}^*}{\gamma(\zeta_1 - \zeta_0)^2} < \beta_1 \le \frac{4}{\gamma(\zeta_1 - \zeta_0)^2},\tag{3.114}$$

and

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 \le \beta_2 \le \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}.$$
 (3.115)

Proof. For this proof, it suffices to find the condition for β_1 such that for any initial condition $x_0 \in \mathcal{A}_2^c$ after a certain number of iterations, the orbit goes into the interval \mathcal{A}_2 . Where $\mathcal{A}_2^c = I \setminus \mathcal{A}_2$ denotes the complement of \mathcal{A}_2 . To conduct the proof, we take the midpoint of the interval $\zeta_0 \leq x < \zeta_1$ as denoted in (3.99) and substitute it into (3.2), where the goal is for this operation to be greater than the fixed point $x_{2,2}^*$ denoted in (3.15), ensuring the existence of monostability. The above can be expressed as:

$$f_{D_1}(x_{m_1}, \gamma, \beta_1) > x_{2,2}^*$$

Performing the substitution, we have

$$-\gamma\beta_1\left[\frac{(\zeta_0+\zeta_1)^2}{4}-\frac{(\zeta_0+\zeta_1)^2}{2}+\zeta_1\zeta_0\right] > x_{2,2}^*.$$

By simplifying the previous inequality

$$\gamma \beta_1 \left[\frac{(\zeta_1 - \zeta_0)^2}{4} \right] > x_{2,2}^*.$$

Finally, by solving for β_1 in the previous expression, we find that

$$\beta_1 > \frac{4x_{2,2}^*}{\gamma(\zeta_1 - \zeta_0)^2},\tag{3.116}$$

which is the lower bound of the interval (3.114). To prove the upper bound, let us note that the maximum point of the unimodal map f_{D_1} is located at $(x, f(x)) = (x_{m_1}, 1)$. Thus, by substituting this maximum point into f_{D_1} , we have that

$$f_{D_1}(x_{m_1}, \gamma, \beta_1) \le 1.$$

By simplifying, we have

$$\gamma \beta_1 \left[\frac{(\zeta_1 - \zeta_0)^2}{4} \right] \le 1.$$

Where, upon solving for β_1 , we obtain

$$\beta_1 \le \frac{4}{\gamma(\zeta_1 - \zeta_0)^2},\tag{3.117}$$

which is the upper bound of the interval (3.114). Finally, by combining the expressions (3.116) and (3.117), we obtain the interval

$$\frac{4x_{2,2}^*}{\gamma(\zeta_1 - \zeta_0)^2} < \beta_1 \le \frac{4}{\gamma(\zeta_1 - \zeta_0)^2}.$$
(3.118)

Notice that the interval for β_2 arises from Proposition 6, where if β_2 satisfies (3.115), then there exists a trapping region or an invariant set \mathcal{A}_2 contained within $[\zeta_1, \zeta_2]$, denoted by (3.79).

The following proposition is a result of the combination of Propositions 1, 6, and 7. It can be observed that the interval (3.119) is the complement of the intervals stated in Propositions 1 and 5. This is done to prevent, after an arbitrarily finite number of iterations, the dynamics from remaining in the second mode. The interval (3.120) corresponds to the one presented in Proposition 7, aiming to maintain the dynamics in the first mode. This is achieved by characterizing an invariant set, as demonstrated in Proposition 7.

Proposition 9 The bimodal map f_D given by (3.1) exhibits monostability in the interval $\zeta_0 \leq x < \zeta_1$ if

$$\beta_2 \in \left[0, \frac{(\sqrt{\zeta_1} + \sqrt{\zeta_2})^2}{\gamma(\zeta_2 - \zeta_1)^2}\right) \bigcup \left(\frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}, \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}\right] \quad (3.119)$$

and

$$\frac{1}{\gamma\zeta_1} \le \beta_1 \le \frac{4}{\gamma\zeta_1}.\tag{3.120}$$

Proof. This proposition arises from Propositions 1,6 and 7. From Proposition 7 states that if

$$\frac{1}{\gamma\zeta_1} \le \beta_1 \le \frac{4}{\gamma\zeta_1},$$

then there exists a trapping region or invariant set denoted as \mathcal{A}_1 . From Proposition 1 we know that there exists fixed points in the subdomain $\zeta_1 \leq x \leq \zeta_2$ when

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2}.$$

By the previous, notice that if

$$0 < \beta_2 < \frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2,$$

then, there not exists fixed points in the subdomain $\zeta_1 \leq x \leq \zeta_2$. Therefore, any initial condition $x_0 \in [0, 1]$, the orbit converges to invariant set or trapping region \mathcal{A}_1 denoted in the Proposition 7. By the Proposition 6, we know that there exists a trapping region or an invariant set when

$$\frac{1}{\gamma} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 \le \beta_2 \le \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2}$$

By the previous, if β_2 takes values within

$$\frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\gamma(\zeta_2 - \zeta_1)^2} < \beta_2 \le \frac{4}{\gamma(\zeta_2 - \zeta_1)^2},$$

then, there not exists an invariant set or a trapping region in $\zeta_1 \leq x \leq \zeta_2$, this means that any initial condition $x_0 \in [0, 1]$ converges to unique invariant set or trapping region \mathcal{A}_1 in the interval I = [0, 1].

In the next chapter, numerical results based on the findings from this chapter are presented. The numerical results cover three cases. In the first case, the parameters γ and β_1 are kept fixed, while $\beta_2 \in (0, 4]$ varies. In the second case, the parameters γ and β_2 remain fixed, while $\beta_1 \in (0, 4]$ varies. Finally, β_1 and β_2 are kept fixed and equal, while $\gamma \in (0, 4]$ varies. In each case, monostability and bistability are studied as the specified parameters change. The results presented include cobweb diagrams, as well as bifurcation diagrams for each case study.

Chapter 4

Families of bimodal maps

In this Chapter 4, we study three parametric families of bimodal maps by applying the propositions given in the previous Chapter 3. The parametric families of bimodal maps present the transition from monostability to bistability and vice versa. Additionally, in each case study, the corresponding bifurcation diagram is presented for the studied family.

4.1 The monoparametric family of maps for the parameter β_2

Below are the assumptions that must be taken into account to generate a monoparametric family of bimodal maps based on the system (3.1) for this case study:

- A regular partition of the interval $I = [0, 1] \subset \mathbb{R}$ given by the set of points $\left\{\zeta_0 = 0, \zeta_1 = \frac{1}{2}, \zeta_2 = 1\right\}$.
- The parameters $\gamma = 4$ and $\beta_1 = 2$.
- The parameter $\beta_2 \in (0, 4] \in \mathbb{R}$.

Under the previous assumptions, a monoparametric family of bimodal maps (3.1) is defined as

$$f_D(x,\beta_2) = 4 \begin{cases} 2x\left(\frac{1}{2} - x\right) & \text{for } 0 \le x < \frac{1}{2}, \\ \beta_2\left(x - \frac{1}{2}\right)(1 - x) & \text{for } \frac{1}{2} \le x \le 1. \end{cases}$$
(4.1)

According to equation (3.98) of the Proposition 7, the system (4.1) presents an invariant set \mathcal{A}_1 when $\beta_1 = \frac{4}{\gamma\zeta_1} = 2$. Based on the previous result, the following proposition emerges.



Figure 4.1: Bimodal maps of the monoparametric family f_D (4.1) for different values of β_2 . The black line corresponds to $\beta_2 = 2$, the red line $\beta_2 = 3$, and finally, the blue one $\beta_2 = 4$.

Proposition 10 The set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right] \subset I \subset \mathbb{R}$ is the invariant set for $f_D(x, \beta_2)$ given by (4.1) when $\beta_1 = 2$.

Proof. For this proof is enough to demonstrate that when the parameter $\beta_1 = 2$ satisfies the invariant set definition, showed in Definition 34. We notice that f_{D_1} has the following form

$$f_{D_1}(x, \gamma = 4, \beta_1 = 2) = 8x \left(\frac{1}{2} - x\right)$$

By deriving the previous expression, we have

$$Df_{D_1}(x) = 4(1 - 4x)$$

If we set $Df_{D_1}(x) = 0$, we obtain the critical value. Then, following the previous procedure, we find that the critical value x_{c_1} is located at $x_{c_1} = \frac{1}{4}$. Notice that $x_{c_1} = x_{m_1}$. Also, notice that the function f_{D_1} is monotonically increasing for $0 \le x < x_{c_1}$ and monotonically decreasing for $x_{c_1} < x < \frac{1}{2}$. Therefore, there exists a maximum critical point at $f_{D_1}(x_{c_1}) = f_{D_1}\left(\frac{1}{4}\right) = \frac{1}{2}$. Also, we have that $f_{D_1}\left(\frac{1}{2}\right) = f_{D_1}(0) = 0$, indicating that $f_{D_1}(0) < f_{D_1}\left(\frac{1}{4}\right)$ and $f_{D_1}\left(\frac{1}{2}\right) < f_{D_1}\left(\frac{1}{4}\right)$. Therefore, by continuity of the function $f_{D_1}(x)$ and also, by the result that when $f_{D_1}(x_{c_1}) = f_{D_1}\left(\frac{1}{4}\right) = \frac{1}{2}$, which is the upper bound of \mathcal{A}_1 , satisfying $f_{D_1}(\mathcal{A}_1) = \mathcal{A}_1$, we can conclude that $\mathcal{A}_1 = \left[0, \frac{1}{2}\right]$ is an invariant set when $\beta_1 = 2$.

According to equation (3.5) of the Proposition 1, we know that the bimodal maps of the monoparametric family given by (4.1) have fixed points in the interval $\frac{1}{2} \leq x \leq 1$, if the parameter $\beta_2 \in \left(\frac{3}{2} + \sqrt{2}, 4\right] \subset \mathbb{R}$. Then the maps of the monoparametric family (4.1) have not fixed points in the interval $\frac{1}{2} \leq x \leq 1$, if the parameter $\beta_2 \in \left(0, \frac{3}{2} + \sqrt{2}\right) \subset \mathbb{R}$.

Proposition 11 The monoparametric family of maps given by (4.1) presents monostability when $\beta_2 \in \left(0, \frac{3}{2} + \sqrt{2}\right)$.

Proof. Graphically, notice that the graph of f_{D_2} is below the identity function when $\beta_2 \in \left(0, \frac{3}{2} + \sqrt{2}\right)$, analytically, if $\beta_2 \in \left(0, \frac{3}{2} + \sqrt{2}\right)$ then the discriminant Δ of the expression for the fixed points showed in (3.8) satisfies $\Delta < 0$ and therefore, have not fixed points in the real numbers field. According to previous aforementioned comments, for every initial condition $x_0 \in \left[\frac{1}{2}, 1\right]$, the orbit $\mathcal{O}(x_0, f_D) = \{x_0, x_1, x_2, \ldots\}$ enters the interval $0 \leq x < \frac{1}{2}$ for a few iterations. Since there is only one invariant set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right)$ then the only possibility for orbits with initial condition $x_0 \in I$ is to converge or remain in \mathcal{A}_1 .

For $\beta_2 = \frac{3}{2} + \sqrt{2}$, the bimodal map has only one fixed point in the interval $\frac{1}{2} \leq x \leq 1$ at $x_2^* = \frac{\sqrt{2}}{2}$, given that when $\beta_2 = \frac{3}{2} + \sqrt{2}$, the discriminant Δ of the expression for the fixed points for f_{D_2} expressed in (3.8) is $\Delta = 0$.

Figure 4.1 shows maps of the monoparametric family (4.1) for different values of $\beta_2 \in \left(\frac{3}{2} + \sqrt{2}, 4\right]$. These maps have fixed points at the intersection between the unimodal map f_{D_2} and the identity function f(x) = x. In this case, the Propositions 6 let us compute a second trapping region or an invariant set $\mathcal{A}_2 \subset \left[\frac{1}{2}, 1\right]$ of the monoparametric family (4.1), the result is:

$$\mathcal{A}_2(\beta_2) = \left[\frac{6\beta_2 - 1 - \sqrt{(6\beta_2 - 1)^2 - 32\beta_2^2}}{8\beta_2}, \frac{6\beta_2 + 1 + \sqrt{(6\beta_2 - 1)^2 - 32\beta_2^2}}{8\beta_2}\right].$$
 (4.2)

A trapping region or an invariant set $\mathcal{A}_2(\beta_2)$ exists if $\frac{3}{2} + \sqrt{2} < \beta_2 \leq \frac{3 + \sqrt{17}}{2}$. Note that \mathcal{A}_2 depends on the parameter β_2 ; thus, we begin conducting the relevant simulations to observe how the bistability occurs.

Proposition 12 The trapping region $\mathcal{A}_2(3) = \left[\frac{2}{3}, \frac{5}{6}\right]$ contains an invariant point at $x_{2,1}^* = \frac{3}{4}$ of the bimodal map of monoparametric family (4.1) for $\beta_2 = 3$.

Proof. For this proof is enough to demonstrate that within the trapping region $\mathcal{A}_2(3)$, there exist two fixed point $x_{2,1}^*$ and $x_{2,2}^*$ denoted by (3.14) and (3.15) respectively, where the fixed point $x_{2,1}^*$ is an attractive fixed point and

therefore, any initial condition $x_0 \in \mathcal{A}_2(3)$ converges to the attractive fixed point $x_{2,1}^*$. From Proposition 1, we know that the bimodal map (4.1) has two fixed points located at $x_{2,2}^* = \frac{2}{3}$ and $x_{2,1}^* = \frac{3}{4}$ for $\beta_2 = 3$, given that this value of β_2 belongs to the interval showed in (3.5). The equation for the unimodal map in the interval $\frac{1}{2} \leq x \leq 1$ is

$$f_{D_2}(x, \gamma = 4, \beta_2 = 3) = 12\left(x - \frac{1}{2}\right)(1 - x),$$

therefore, by deriving the previous expression

$$Df_{D_2}(x) = -6(4x - 3), (4.3)$$

applying the Definition 24 for the fixed point $x_{2,2}^*$ substituting into (4.3) we have that

$$\left| Df_{D_2}(x_{2,2}^*) \right| = \left| -6\left(4\left(\frac{2}{3}\right) - 3 \right) \right| = |2| > 1,$$

therefore, the fixed point $x_{2,2}^*$ is a repulsive fixed point. We doing the same procedure for the fixed point $x_{2,1}^*$ and it results

$$\left| Df_{D_2}(x_{2,1}^*) \right| = \left| -6\left(4\left(\frac{3}{4}\right) - 3 \right) \right| = 0 < 1.$$

Therefore, the fixed point $x_{2,1}^*$ is an attractive fixed point. By the previous results, we can state that the former is a repulsive fixed point and the second is an attracting fixed point $x_{2,1}^* = \frac{3}{4}$ that fulfills the conditions of Proposition 3. We know that the map (4.1) is a unimodal map in the interval $x \in \begin{bmatrix} \frac{2}{3}, \frac{5}{6} \end{bmatrix}$ and has a critical value at $x_{c_2} = \frac{3}{4}$ (notice that $x_{c_2} = x_{m_2}$) given that if we set $Df_{D_2}(x) = 0$ we find that the solution that satisfies the $Df_{D_2}(x) = 0$ is $x_{c_2} = \frac{3}{4}$. Also, the map is monotonically increasing at $\frac{2}{3} \leq x < x_{c_2}$ and monotonically decreasing at $x_{c_2} < x \leq \frac{5}{6}$ and there exists a maximum critical point at $f_{D_2}(x_c) = f_{D_2}\left(\frac{3}{4}\right) = \frac{3}{4}$. Analogously, by the previous, we have that $f_D\left(\frac{3}{4}\right) = \frac{3}{4}$, and $f_D\left(\frac{2}{3}\right) = f_D\left(\frac{5}{6}\right) = \frac{2}{3}$, such that

 $f_D\left(\frac{2}{3}\right) < f_D\left(\frac{3}{4}\right)$ and $f_D\left(\frac{5}{6}\right) < f_D\left(\frac{3}{4}\right)$, then for continuity the function $f_D(x_i): \left[\frac{2}{3}, \frac{5}{6}\right] \rightarrow \left[\frac{2}{3}, \frac{5}{6}\right]$. As $x_{2,1}^* = \frac{3}{4}$ is an attracting point then any initial condition $x_0 \in \left(\frac{2}{3}, \frac{5}{6}\right)$ converges to $x_{2,1}^* = \frac{3}{4}$.

Figures 4.2 and 4.3 show the orbit for $x_0 = 0.68$ and we can see how the orbit converges to the fixed point $x_{2,1}^* = 0.75$, while $x_{2,2}^*$ is a repulsive fixed point according to Proposition 2. Figure 4.4 shows the orbit of an initial condition $x_0 \notin \mathcal{A}_2(3)$, after a certain number of iterations, the dynamics enters to the invariant set \mathcal{A}_1 . Now, we consider a new value of $\beta_2 = 3.3 \in \left[\frac{3}{2} + \sqrt{2}, \frac{3 + \sqrt{17}}{2}\right] \subset \mathbb{R}$ and its corresponding trapping region is $\mathcal{A}_2(3.3) = \left[\frac{47 - \sqrt{31}}{66}, \frac{52 + \sqrt{31}}{66}\right] \subset \mathbb{R}$. Figures 4.5 and 4.6 shows the dynamics of f_D in which we notice that the orbit remains within $\mathcal{A}_2(3.3)$

indicating bistability. Furthermore, the period-two orbit is exhibited.



Figure 4.2: Cobweb diagram for the bimodal map f_D with $\beta_2 = 3$ using the initial condition $x_0 = 0.68 \in \mathcal{A}_2(3)$.


Figure 4.3: Orbit of the bimodal map f_D with $\beta_2 = 3$ using the initial condition $x_0 = 0.68 \in \mathcal{A}_2(3)$, where it's possible to observe the convergence of the trajectory towards the attractor fixed point $x_{2,1}^* = \frac{3}{4}$.



Figure 4.4: Cobweb diagram for the bimodal map f_D with $\beta_2 = 3$ with initial condition $x_0 = 0.85$ such that $x_0 \notin \mathcal{A}_2(3)$.



Figure 4.5: Cobweb diagram for the bimodal map f_D with $\beta_2 = 3.3$ using different initial conditions $x_0 \in \mathcal{A}_2(3.3)$; $x_0 = 0.87$.



Figure 4.6: Orbit of the bimodal map f_D with $\beta_2 = 3.3$ using the initial condition $x_0 \in \mathcal{A}_2(3.3)$; $x_0 = 0.87$, where it's possible to observe a period-two orbit.

Proposition 13 The trapping region $\mathcal{A}_2(3.3) = \left[\frac{47 - \sqrt{31}}{66}, \frac{52 + \sqrt{31}}{66}\right] \subset \mathbb{R}$ contains attracting periodic orbit $x_1 = \frac{52 + \sqrt{6}}{66}$ and $x_2 = \frac{52 - \sqrt{6}}{66}$ of the bimodal map of monoparametric family (4.1) for $\beta_2 = 3.3$.

Proof. For this proof, we utilize the result obtained from Proposition 5. We know that the period-doubling bifurcation occurs when β_2 satisfies (3.49), specifically when $\beta_2 = \frac{3}{2} + \sqrt{3}$. Furthermore, the period-two points defined in (3.57) and (3.58) satisfy $x_1 > x_2$ when $\beta_2 > \frac{3}{2} + \sqrt{3}$. For the trapping region $\mathcal{A}_2(3.3)$ defined above, the value of the parameter β_2 is set to $\beta_2 = 3.3$, which satisfies $3.3 > \frac{3}{2} + \sqrt{3}$. Therefore, based on Proposition 5, we conclude that the period-two points are $x_1 = \frac{52 + \sqrt{6}}{66}$ and $x_2 = \frac{52 - \sqrt{6}}{66}$. To demonstrate whether this periodic orbit is attracting, it is sufficient to show that the following condition is fulfilled:

$$|Df_{D_2}(x_1)||Df_{D_2}(x_2)| < 1.$$

Here

$$f_{D_2}(x, \gamma = 4, \beta_2 = 3.3) = 13.2\left(x - \frac{1}{2}\right)(1 - x),$$

and therefore, their derivative

$$Df_{D_2}(x) = -26.4x + 19.8.$$

Applying the previous condition, we have:

$$|Df_{D_2}(x_1)||Df_{D_2}(x_2)| = \left|\frac{1}{25}\right| < 1.$$

Hence, the periodic orbit is attracting.

Taking a new value of $\beta_2 = \frac{3 + \sqrt{17}}{2}$, its respective invariant set is $\mathcal{A}_2\left(\frac{3 + \sqrt{17}}{2}\right) = \left[\frac{9 - \sqrt{17}}{8}, \frac{3 + \sqrt{17}}{8}\right] \subset \mathbb{R}$. In Figure 4.7, we can observe

the dynamics of f_D for initial conditions $x_0 = 0.61 \in \mathcal{A}_2\left(\frac{3+\sqrt{17}}{2}\right)$, where we notice that the orbit remains within the invariant set $\mathcal{A}_2\left(\frac{3+\sqrt{17}}{2}\right)$, giving rise to bistability.



Figure 4.7: Cobweb diagram for the bimodal map f_D with $\beta_2 = \frac{3 + \sqrt{17}}{2}$ using the initial condition $x_0 = 0.61 \in \mathcal{A}_2\left(\frac{3 + \sqrt{17}}{2}\right)$.

Proposition 14 $\mathcal{A}_2\left(\frac{3+\sqrt{17}}{2}\right) = \left[\frac{9-\sqrt{17}}{8}, \frac{3+\sqrt{17}}{8}\right] \subset \mathbb{R}$ is an invari-

ant set of the bimodal map of monoparametric family (4.1) for $\beta_2 = \frac{3 + \sqrt{17}}{2}$.

Proposition 15 There exist two invariant sets $A_1 = \begin{bmatrix} 0, \frac{1}{2} \end{bmatrix}$ and $A_2 \left(\frac{3 + \sqrt{17}}{2} \right) = \begin{bmatrix} \frac{9 - \sqrt{17}}{8}, \frac{3 + \sqrt{17}}{8} \end{bmatrix}$ of the bimodal map of monoparametric family (4.1) for $\beta_2 = \frac{3 + \sqrt{17}}{2}$ then the system presents bistability.

Finally, the last case is when $\beta_2 = 3.57 \in \left(\frac{3+\sqrt{17}}{2}, 4\right]$, for this realization its corresponding set $\mathcal{A}_2(3.57) = \left[\frac{1021-\sqrt{22849}}{1428}, \frac{1121+\sqrt{22849}}{1428}\right] \subset \mathbb{R}$ is not an invariant set neither a trapping region, then bistability disap

 \mathbb{R} is not an invariant set neither a trapping region, then bistability disappears. Every orbit $x_0 \in \mathcal{A}_2(3.57)$ converges to the invariant set \mathcal{A}_1 . In Figure 4.8, we can observe the dynamics of the bimodal map f_D with $x_0 \in \mathcal{A}_2(3.57)$.



Figure 4.8: Cobweb diagram for the bimodal map f_D with $\beta_2 = 3.57$ using the initial condition $x_0 = 0.61 \in \mathcal{A}_2(3.57)$.

With the previous results, it can be verified that indeed, the conditions provided in the propositions presented in Chapter 3 work to generate bistability in the bimodal map f_{D_2} as presented in equation (3.1). Particularly, we have studied bistability in the monoparametric family of bimodal maps (4.1) when $\beta_2 \in \left[\frac{3}{2} + \sqrt{2}, \frac{3 + \sqrt{17}}{2}\right]$, and other case, the system (4.1) presents monostability. The bifurcation diagram for the parameter $\beta_2 \in [0, 4]$ is shown in Figure 4.9. Below we briefly describe the different behaviors observe in the monoparametric family of multimodal maps given by (4.1).



Figure 4.9: Bifurcation diagram of the bimodal map f_D given by (4.1) with respect to the parameters β_1 and β_2 , where $\beta_1 = 2$ and $\beta_2 \in (0, 4] \subset \mathbb{R}$.

- For $\beta_2 \in \left[0, \frac{3}{2} + \sqrt{2}\right)$, the maps presents two fixed points in the interval $0 \leq x < \frac{1}{2}$ and none in the interval $\frac{1}{2} \leq x \leq 1$. Every orbit $x_0 \in [0, 1]$ finally oscillates in the invariant set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right)$, then the family of maps exhibits a monostable behavior.
- For $\beta_2 = \frac{3}{2} + \sqrt{2}$, the family presents a tangent bifurcation and the system presents two fixed points in the interval $0 \le x < \frac{1}{2}$ and one in the interval $\frac{1}{2} \le x \le 1$. The map exhibits a monostable behavior and every orbit $x_0 \in [0, 1]$ converges to the invariant set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right)$.
- For $\beta_2 \in \left(\frac{3}{2} + \sqrt{2}, \frac{3}{2} + \sqrt{3}\right)$, the family of maps presents two fixed points in the interval $0 \le x < \frac{1}{2}$ and two in the interval $\frac{1}{2} \le x \le 1$. The map exhibits a bistable behavior and every orbit $x_0 \in [0, 1]$ converges

to the invariant set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right)$ or belongs to the trapping region $\mathcal{A}_2(\beta_2)$ that contains a attracting fixed point.

- For $\beta_2 = \frac{3}{2} + \sqrt{3}$, the family presents a period-doubling bifurcation.
- For $\beta_2 \in \left(\frac{3}{2} + \sqrt{3}, \frac{3 + \sqrt{17}}{2}\right]$, the family presents four repelling fixed points and at the beginning the bistability is generated by the trapping region $\mathcal{A}_2(\beta_2)$ and the invariant set \mathcal{A}_1 . Finally, the bistability is generated by two invariant sets \mathcal{A}_1 and $\mathcal{A}_2(\beta_2)$ when $\beta_2 = \frac{3 + \sqrt{17}}{2}$.
- For $\beta_2 \in \left(\frac{3+\sqrt{17}}{2}, 4\right]$, the family presents monostability again in the interval $0 \le x < \frac{1}{2}$.

From the previous numerical results of this study cases, we can observe that any initial condition $x_0 \in [\zeta_1, \zeta_2] \setminus \mathcal{A}_2(\beta_2)$ converges to the invariant set \mathcal{A}_1 within the interval $\zeta_0 \leq x < \zeta_1$. Therefore, there exists a basin of attraction in $x \in [\zeta_1, \zeta_2] \setminus \mathcal{A}_2(\beta_2) \subset [0, 1]$ and an attractor in \mathcal{A}_1 for any $\beta_2 \in (0, 4] \subset \mathbb{R}$.

4.2 The monoparametric family of maps for the parameter β_1

The assumptions taken into account to generate a monoparametric family of bimodal maps based on the system (3.1) are as follows:

- A regular partition of the interval I = [0, 1] given by the set of points $\{\zeta_0 = 0, \zeta_1 = \frac{1}{2}, \zeta_2 = 1\}.$
- The parameters $\gamma = 4$ and $\beta_2 = 3.56$ is chosen to ensure the existence of the trapping region $\mathcal{A}_2(\beta_2)$ in the subdomain $\zeta_1 \leq x \leq \zeta_2$, thus ensuring monostability and bistability.

• The parameter $\beta_1 \in (0, 4] \subset \mathbb{R}$.

Under the previous assumptions, a monoparametric family of bimodal maps (3.1) is defined as

$$f_D(x,\beta_1) = 4 \begin{cases} \beta_1 x \left(\frac{1}{2} - x\right) & \text{for } 0 \le x < \frac{1}{2}, \\ 3.56 \left(x - \frac{1}{2}\right) (1 - x) & \text{for } \frac{1}{2} \le x \le 1. \end{cases}$$
(4.4)

Figure 4.15 displays the graph of system (4.4) for different values of β_1 : The red line represents $\beta_1 = 1$, the black line $\beta_1 = 2$, and finally, the pink one $\beta_1 = 4$. Considering the map (4.4), we proceed to study bistability and monostability.



Figure 4.10: Bimodal map f_D (4.4) with $\gamma = 4$, $\beta_2 = 3.56$ and different values of β_1 . The black line represents $\beta_1 = 2$, the red line $\beta_1 = 3$, and finally, the blue one $\beta_1 = 4$.

Proposition 16 The system (4.4) has a trapping region $\mathcal{A}_1 = \left[0, \frac{1}{2}\right] \subset \mathbb{R}$ where all trajectories $x_0 \in \mathcal{A}_1$ converges to $x_{1,1}^* = \frac{2\beta_1 - 1}{4\beta_1}$ or $x_{1,2}^* = 0$ if

$$\beta_1 \in \left(0, \frac{3}{2}\right).$$

Proof. For this proof, it's necessary to show that when $\beta_1 \in \left(0, \frac{3}{2}\right)$ there exists a trapping region $\mathcal{A}_1 = \left[0, \frac{1}{2}\right] \subset \mathbb{R}$ and additionally, we need proof that the fixed points $x_{1,1}^*$ and $x_{1,2}^*$ are attractive when β_1 takes the values previously mentioned. We proceed to calculate the critical point $x_{c_1} \in \mathcal{A}_1$ for the first modal. For this, we calculate the derivative of f_{D_1} , getting

$$Df_{D_1}(x) = 2\beta_1(1 - 4x).$$

By setting $Df_{D_1} = 0$ we have

$$2\beta_1(1-4x) = 0.$$

Since $\beta_1 > 0$, we have

$$1 - 4x = 0,$$

by solving the previous equation, we obtain that the critical point of the first modal located at $x_{c_1} = \frac{1}{4} \in \mathcal{A}_1$. Now, we defined the interval $I_0 = [0, x_c]$, $I_1 = \left[x_c, \frac{1}{2}\right)$, such that $\mathcal{A}_1 = I_0 \cup I_1$. Therefore, notice that

$$f_{D_1}(I_0,\beta_1) = f_{D_1}(I_1,\beta_1) = \left[0,\frac{3}{8}\right) \quad \forall \beta_1 \in \left(0,\frac{3}{2}\right).$$

By the previous, notice that we can conclude that

$$f_{D_1}(I_0 \cup I_1, \beta_1) = f_{D_1}(\mathcal{A}_1, \beta_1) = \left[0, \frac{3}{8}\right] \subset \mathcal{A}_1 \quad \forall \beta_1 \in \left(0, \frac{3}{2}\right).$$

Therefore, the orbit of each initial condition $x_0 \in \mathcal{A}_1$ remains in \mathcal{A}_1 , i.e., $f(\mathcal{A}_1) \subset \mathcal{A}_1$ that is the definition of trapping region. Now, we proceed to show that one of the fixed points $x_{1,1}^*$ or $x_{1,2}^*$ is attractive when $0 < \beta_1 < \frac{3}{2}$. We know that the equation for the first modal is $f_{D_1}(x,\beta_1) = 4\beta_1 x \left(\frac{1}{2} - x\right)$

and their respective derivative is $Df_{D_1}(x,\beta_1) = 2\beta_1(1-4x)$. Now, we find the condition on β_1 such that

$$|Df_{D_1}(x_{1,n}^*)| < 1 \text{ for } n = \{1,2\}.$$

For the fixed point $x_{1,1}^*$, we have

$$|Df_{D_1}(x_{1,1}^*)| = |2(\beta_1 - 1)| < 1,$$

analogously we have

$$-1 < 2(\beta_1 - 1) < 1$$

By solving the previous inequality for β_1 , we conclude that when $\frac{1}{2} < \beta_1 < \frac{3}{2}$, the fixed point $x_{1,1}^*$ is attractive. Performing the same procedure for the fixed point $x_{1,2}^*$, we have

$$|Df_{D_1}(x_{1,2}^*)| = |2\beta_1| < 1.$$

Again, by solving the previous expression for β_1 , we conclude that when $0 < \beta_1 < \frac{1}{2}$, the fixed point $x_{1,2}^*$ is attractive. Combining the two intervals for β_1 found previously, we conclude that if $0 < \beta_1 < \frac{3}{2}$, then all trajectories $x_0 \in \mathcal{A}_1$ converge to $x_{1,1}^*$ or $x_{1,2}^*$.

Proposition 17 The system (4.4) has an invariant set $\mathcal{A}_1 = \left[0, \frac{1}{2}\right] \subset \mathbb{R}$ if $\beta_1 = 2$.

Proof. This proof is similar to the previous one. For this proof it's necessary to show that when $\beta_1 = 2$ there exists an invariant set $\mathcal{A}_1 = \begin{bmatrix} 0, \frac{1}{2} \end{bmatrix} \subset \mathbb{R}$. We know from the proof of the Proposition 16 that the point $x_{c_1} \in \mathcal{A}_1$ is the critical point for the first modal. We proceed to state the following intervals which were mentioned previously: $I_0 = [0, x_c], I_1 = \begin{bmatrix} x_c, \frac{1}{2} \end{bmatrix}$, and we remember that these intervals satisfy $\mathcal{A}_1 = I_0 \cup I_1$. Therefore, notice that

$$f_{D_1}(I_0, \beta_1 = 2) = f_{D_1}(I_1, \beta_1 = 2) = \left[0, \frac{1}{2}\right).$$

Based on the above, we can conclude that

$$f_{D_1}(I_0 \cup I_1, \beta_1 = 2) = f_{D_1}(\mathcal{A}_1, \beta_1 = 2) = \left[0, \frac{1}{2}\right] = \mathcal{A}_1.$$

Notice that the previous result satisfies the definition of invariant set. Therefore, for any orbit $x_0 \in A_1$ it remains in A_1 and moreover, completely covers A_1 .

Notice that when $\beta_1 = 2$, the system (4.4) has an invariant set \mathcal{A}_1 in the subdomain $\zeta_0 \leq x < \zeta_1$ and a trapping region $\mathcal{A}_2(\beta_2 = 3.56)$ within $\zeta_1 \leq x \leq \zeta_2$. Only the invariant set \mathcal{A}_1 has a basin of attraction given by $I \setminus \mathcal{A}_1(\beta = 2) \cup \mathcal{A}_2(\beta_1 = 3.56)$. Figures 4.11 and 4.12 show the cobweb diagram and their respective trajectories diagram for the bimodal map f_D with $\beta_1 = 2$ and $\beta_2 = 3.56$ using initial conditions, where one of them belongs to the invariant set \mathcal{A}_1 and the other belongs to trapping region $\mathcal{A}_2(3.56)$.



Figure 4.11: (a) Cobweb diagram and (b) orbit of the bimodal map f_D with $\beta_1 = 2, \beta_2 = 3.56$ and $x_0 = 0.0357 \in \mathcal{A}_1$.

If we consider Proposition 8, we know that when $\frac{3}{2} + \sqrt{3} \leq \beta_2 \leq \frac{3 + \sqrt{17}}{2}$ and $4x_{2,2}^* < \beta_1 \leq 4$, monostability exists in the interval $\zeta_1 \leq x \leq \zeta_2$. Thus, taking $\beta_1 = 2.5$, we note that regardless of the initial condition, after a finite and small number of iterations, the orbit will eventually converge to the invariant set $\mathcal{A}_2(3.56)$. For this case, the invariant set \mathcal{A}_2 has a basin of attraction given by $I \setminus \mathcal{A}_2(3.56)$. Figure 4.13 shows the cobweb diagram for the bimodal map f_D with $\beta_1 = 2.5$ using an initial condition in the basin of attraction $x_0 = 0.9502 \notin \mathcal{A}_2(3.56)$.



Figure 4.12: (a) Cobweb diagram and (b) orbit of the bimodal map f_D with $\beta_1 = 2, \beta_2 = 3.56$ and $x_0 = 0.6557 \in \mathcal{A}_2(3.56)$.



Figure 4.13: (a) Cobweb diagram and (b) orbit of the bimodal map f_D with $\beta_1 = 2.5$ and $\beta_2 = 3.56$ using the initial condition $x_0 = 0.9502 \notin \mathcal{A}_2(3.56)$.

The bifurcation diagram based on the behavior presented in this case study was constructed and is shown in Figure 4.14, in which we can observe that the behaviors in the previously presented cobweb diagram are related to this bifurcation diagram. Below we briefly describe the different behaviors observe in the monoparametric family of multimodals maps given by (4.4).

- For $\beta_1 \in \left(0, \frac{1}{2}\right)$ and $\beta_1 \in \left(\frac{1}{2}, 4\right]$ the fixed point $x_{1,2}^* = 0$ is attractive and repulsive, respectively.
- For $\beta_1 \in \left(\frac{1}{2}, \frac{3}{2}\right)$ and $\beta_1 \in \left(\frac{3}{2}, 4\right]$ the fixed point $x_{1,1}^* = \frac{2\beta_1 1}{4\beta_1}$ is attractive and repulsive, respectively.

- For $\beta_1 = \frac{3}{2}$, the family presents a period-doubling bifurcation.
- For $\beta_1 \in (0, 2)$ the family of bimodal maps (4.4) presents two trapping regions $\mathcal{A}_1 = \begin{bmatrix} 0, \frac{1}{2} \end{bmatrix}$ and $\mathcal{A}_2(3.56) \subset \begin{bmatrix} \frac{1}{2}, 1 \end{bmatrix}$, therefore the maps exhibit a bistable behavior and every orbit $x_0 \notin \mathcal{A}_2(3.56)$ converges or belongs to the trapping region $\mathcal{A}_1 = \begin{bmatrix} 0, \frac{1}{2} \end{bmatrix}$.
- For $\beta_1 = 2$, the map (4.4) exhibits a bistable behavior via an invariant set $\mathcal{A}_1 = [0, 0.5)$ and a trapping region $\mathcal{A}_2(3.56)$.
- The family of bimodal maps (4.4) presents monostability in $\zeta_1 \leq x \leq \zeta_2$ when

$$4x_{2,2}^* < \beta_1 \le 4,$$

and

$$\frac{3}{2} + \sqrt{3} \le \beta_2 \le \frac{3 + \sqrt{17}}{2},$$

where $x_{2,2}^*$ is the fixed point of the subdomain $[\zeta_1, \zeta_2]$, denoted by (3.15).



Figure 4.14: Bifurcation diagram of the bimodal map f_D given by (4.4) with respect to the parameters β_1 and β_2 , where $\beta_2 = 3.56$ and $\beta_1 \in (0, 4]$.

4.3 The monoparametric family of maps for the parameter γ

The assumptions taken into account to generate a monoparametric family of bimodal maps based on the system (3.1) are as follows:

- A regular partition of the interval [0, 1] given by the set of points $\{\zeta_0 = 0, \zeta_1 = \frac{1}{2}, \zeta_2 = 1\}.$
- The parameters $\beta_1 = \beta_2 = 4$.
- The parameter $\gamma \in (0, 4] \subset \mathbb{R}$.

Under the previous assumptions, a monoparametric family of bimodal maps (3.1) is defined as



Figure 4.15: Bimodal map f_D (4.5) with $\gamma = 4$, $\beta_1 = \beta_2 = 4$ and different values of $\gamma \in (0, 4]$. The black line represents $\gamma = 2$, the red line $\gamma = 3$, and finally, the blue one $\gamma = 4$.

$$f_D(x,\gamma) = \gamma \begin{cases} 4x \left(\frac{1}{2} - x\right) & \text{for } 0 \le x < \frac{1}{2}, \\ 4\left(x - \frac{1}{2}\right)(1 - x) & \text{for } \frac{1}{2} \le x \le 1. \end{cases}$$
(4.5)

We can state briefly the behavior for the fixed points when the parameter γ adopts specifically certain values.

- when $\gamma \in \left(0, \frac{1}{2}\right)$ the fixed point $x_{1,2}^*$ denoted by (3.110), is an invariant point, moreover, this fixed points is unique within I.
- When $\gamma \in \left(\frac{1}{2}, \frac{3}{2}\right)$ the fixed point $x_{1,2}^*$ is repulsive, and a new invariant appears, this invariant is the fixed point $x_{1,1}^*$ defined by (3.109).
- When $\gamma = \frac{3}{2}$ a period-doubling bifurcations occurs.

Proposition 18 The bimodal map (4.5) exhibits monostability and an invariant set $\mathcal{A} = \left[0, \frac{\gamma}{4}\right] \subset I$ when $\frac{1}{2} \leq \frac{\gamma}{4} < \frac{3+2\sqrt{2}}{8}$.

Proof. For this proof, it's necessary to show that within of the interval $I = [0, 1] \subset \mathbb{R}$ there exists an invariant set when the parameter $\gamma \in \left[2, \frac{3+2\sqrt{2}}{2}\right]$, also, we need proof that f_D exhibits monostability. For the first part of this proof, we show that exist an invariant set, for this, let us consider the following three intervals $I_0 = [0, x_{c_1}], I_1 = \left[x_{c_1}, \frac{1}{2}\right]$ and $I_2 = \left[\frac{1}{2}, \frac{\gamma}{4}\right]$, where x_{c_1} is the critical point belongs to subdomain $0 \le x < \frac{1}{2}$, where $x_{c_1} = \frac{1}{4}$. Notice that the set $\mathcal{A} = I_0 \cup I_1 \cup I_2$.

If we take any $x \in \left[0, \frac{1}{2}\right)$ then the expression for the respective modal is $f_{D_1}(x) = 4\gamma x \left(\frac{1}{2} - x\right)$. By evaluating I_0 and I_1 in f_{D_1} we have the following results: $f_{D_1}(I_0) = \left[0, \frac{\gamma}{4}\right]$, $f_{D_1}(I_1) = \left[0, \frac{\gamma}{4}\right]$. Now, if we take any $x \in \left[\frac{1}{2}, 1\right]$ then their respective modal is given by $f_{D_2}(x) = 4\gamma \left(x - \frac{1}{2}\right)(1 - x)$. Notice that interval I_2 belongs to the subdomain defined by $\frac{1}{2} \leq x \leq 1$ therefore, by evaluating I_2 in f_{D_2} we have $f_{D_2}(I_2) = \left[0, \frac{\gamma}{4}(\gamma - 2)(4 - \gamma)\right]$. We take the upper bound of the parameter γ in this case $\gamma = \frac{3 + 2\sqrt{2}}{2}$. We proceed to substituting in $f_{D_2}(I_2)$, where we obtain

$$\frac{\gamma}{4}(\gamma-2)(4-\gamma) = \left(3\sqrt{2} - \frac{13}{4}\right)\left(\frac{\gamma}{4}\right) < \frac{\gamma}{4}.$$

And as the upper bound of the set \mathcal{A} is not reached by the second modal when $\gamma = \frac{3+2\sqrt{2}}{2}$, then, we conclude that $f(\mathcal{A}) = \mathcal{A}$. Therefore, \mathcal{A} is an invariant set. To show monostability, notice that f_{D_2} has not fixed point. Therefore, for any initial condition $x_0 \in I \setminus \mathcal{A}$ the orbit converges to \mathcal{A} . Therefore, the bimodal map (4.5) presents monostability in the subdomain $0 \leq x \leq \frac{\gamma}{4}$ when

$$2 \le \gamma < \frac{3 + 2\sqrt{2}}{2}.$$



Figure 4.16: Cobweb diagram for the bimodal map f_D for: a) $\gamma = 2$ using the initial condition $x_0 = 0.9157$, and b) $\gamma = 2.5$ using the initial condition $x_0 = 0.7922$.

Figures 4.16 shows the Cobweb diagram for the bimodal map f_D for: a) $\gamma = 2$ using the initial conditions $x_0 = 0.9157$, and b) $\gamma = 2.5$ using the initial conditions $x_0 = 0.7922$. It is possible to see that the bimodal map (4.5) exhibits monostability in the invariant set $\mathcal{A} = \left[0, \frac{\gamma}{4}\right]$ for all $x_0 \in I$.

• If $\frac{3+2\sqrt{2}}{2} \le \gamma < \frac{3+2\sqrt{3}}{2}$, there exists an invariant point denoted $x_{2,1}^*$ defined by (3.14), and at $\gamma = \frac{3+2\sqrt{3}}{2}$ a period-doubling bifurcations occurs.

Proposition 19 If
$$\gamma \in \left(\frac{3+2\sqrt{2}}{2}, \frac{3+\sqrt{17}}{2}\right)$$
 then the system (4.5) has a trapping region $\mathcal{A}(\gamma) \subset \left[\frac{1}{2}, 1\right] \subset \mathbb{R}$ defined as
$$\mathcal{A}(\gamma) = \left[\frac{6\gamma - 1 - \sqrt{(6\gamma - 1)^2 - 32\gamma^2}}{8\gamma}, \frac{6\gamma + 1 + \sqrt{(6\gamma - 1)^2 - 32\gamma^2}}{8\gamma}\right].$$

Proof. To demonstrate this proposition, we will find the value for the parameter γ such that $f_{D_2}^2(x_{c_2}, \gamma) > x_{2,2}^*$, with $x_{c_2} = \frac{\zeta_1 + \zeta_2}{2} \in [\zeta_1, \zeta_2]$ as the critical point of the subdomain $\frac{1}{2} \leq x \leq 1$. The previous condition arises from the definition of the invariant set. Now, we defined the following r, a, b, and c such that $r = \gamma \beta_2$, $a = \zeta_1 + \zeta_2$, $b = \zeta_1 \zeta_2$ and $c = a^4 - 4b$. Therefore, by the previous definitions, notice that x_{c_1} and $x_{2,2}^*$ results

$$x_{c_1} = \frac{a}{2}, \quad x_{2,2}^* = \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}$$

Now, we proceed to solving

$$f_{D_2}^2(x_{c_2},\gamma) > x_{2,2}^*$$

We know that

$$f_{D_2}(x,\gamma,\beta_2) = -\gamma\beta_2[x^2 - (\zeta_1 + \zeta_2)x + \zeta_1\zeta_2] = -rx^2 + rax - rb,$$

by substituting x_{c_2} into the last expression and by simplifying, we obtain

$$f_{D_2}(x_{c_2}, r, a, b) = r \frac{(a^2 - 4b)}{4}$$

when performing $f_{D_2}(f_{D_2}(x_{c_2}, r, a, b)) = f_{D_2}^2(x_{c_2}, r, a, b)$, we have

$$f_{D_2}^2(x_{m_2}, r, a, b) = -r\left(r\frac{(a^2 - 4b)}{4}\right)^2 + ra\left(r\frac{(a^2 - 4b)}{4}\right) - rb,$$

since $c = a^4 - 4b$ then it results

$$f_{D_2}^2(x_{c_2}, r, a, b) = \frac{-r^3c^2 + 4r^2ac - 16rb}{16}$$

Therefore, if we make $f_{D_2}^2(x_{c_2}, r, a, b) > x_{2,2}^*$ we obtain the following result

$$\frac{-r^3c^2 + 4r^2ac - 16rb}{16} < \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}$$

By simplifying and factoring the previous inequality, the result is:

$$(cr^{2} - 2ar - 8)(c^{3}r^{3} - 6ac^{2}r^{2} + 4c(8b + a^{2} + 2c)r - 8a(2c + 8b - a^{2})) < 0.$$

We take the quadratic factor $(cr^2 - 2ar - 8) < 0$ and solving for r, where we obtain the following:

$$r = \frac{2a \pm \sqrt{4a^2 + 32c}}{2c} = \frac{a \pm \sqrt{a^2 + 8c}}{c}$$

By substituting $a = (\zeta_1 + \zeta_2)$, $b = \zeta_1 \zeta_2$, $c = a^4 - 4b$ and $r = \gamma \beta_2$ and by simplifying, we have

$$\gamma\beta_2 = \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{(\zeta_2 - \zeta_1)^2}.$$

Analogously

$$\gamma = \frac{(\zeta_1 + \zeta_2) \pm \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2}$$

Then, the solution for the inequality $(cr^2 - 2ar - 8) < 0$ is

$$\frac{(\zeta_1+\zeta_2)-\sqrt{9(\zeta_1+\zeta_2)^2-32\zeta_1\zeta_2}}{\beta_2(\zeta_2-\zeta_1)^2} < \gamma < \frac{(\zeta_1+\zeta_2)+\sqrt{9(\zeta_1+\zeta_2)^2-32\zeta_1\zeta_2}}{\beta_2(\zeta_2-\zeta_1)^2}.$$

However, notice that the lower bound satisfies that

$$\gamma = \frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2} < 0,$$

also, from Proposition 1 we know that there exists fixed points in f_{D_2} , where

$$\gamma > \frac{1}{\beta_2} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2.$$

Then, notice that

$$\gamma = \frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2} < 0 < \frac{1}{\beta_2} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1}\right)^2.$$

Therefore, the solution set for the inequality $cr^2 - 2ar - 8 < 0$ is

$$\frac{1}{\beta_2} \left(\frac{\sqrt{\zeta_1} + \sqrt{\zeta_2}}{\zeta_2 - \zeta_1} \right)^2 < \gamma < \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2},$$

and by substituting $\beta_2 = 4$, $\zeta_1 = \frac{1}{2}$ and $\zeta_2 = 1$, we obtain the following interval

$$\frac{3+2\sqrt{2}}{2} < \gamma < \frac{3+\sqrt{17}}{2},$$

that which is the solution set that the parameter γ must take for $\mathcal{A}(\gamma)$ to be an invariant set.

Proposition 20 If $\gamma = \frac{3 + \sqrt{17}}{2}$ then the system (4.5) has an invariant set $\mathcal{A}(\gamma) \subset \left[\frac{1}{2}, 1\right] \subset \mathbb{R}$ defined as $\mathcal{A}(\gamma) = \left[\frac{6\gamma - 1 - \sqrt{(6\gamma - 1)^2 - 32\gamma^2}}{8\gamma}, \frac{6\gamma + 1 + \sqrt{(6\gamma - 1)^2 - 32\gamma^2}}{8\gamma}\right].$

Proof. This proof is similar to the previous one. To demonstrate this proposition, we will find the value for the parameter γ such that $f_{D_2}^2(x_{c_2}, \gamma) = x_{2,2}^*$, with $x_{c_2} = \frac{\zeta_1 + \zeta_2}{2} \in [\zeta_1, \zeta_2]$ as the critical point of the subdomain $\frac{1}{2} \leq x \leq 1$. The previous condition arises from the definition of the invariant set. Now, we defined the following parameters: $r = \gamma \beta_2$, $a = \zeta_1 + \zeta_2$, $b = \zeta_1 \zeta_2$ and $c = a^4 - 4b$. Therefore, by the previous definitions, notice that x_{c_1} and $x_{2,2}^*$ results

$$x_{c_1} = \frac{a}{2}$$
 and $x_{2,2}^* = \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}$.

Now, we proceed to solving

$$f_{D_2}^2(x_{c_2},\gamma) = x_{2,2}^*.$$

We know that

$$f_{D_2}(x,\gamma,\beta_2) = -\gamma\beta_2[x^2 - (\zeta_1 + \zeta_2)x + \zeta_1\zeta_2] = -rx^2 + rax - rb.$$

By substituting x_{c_2} into the last expression and by simplifying, we have

$$f_{D_2}(x_{c_2}, r, a, b) = r \frac{(a^2 - 4b)}{4}$$

When performing $f_{D_2}(f_{D_2}(x_{c_2}, r, a, b)) = f_{D_2}^2(x_{c_2}, r, a, b)$. We have

$$f_{D_2}^2(x_{m_2}, r, a, b) = -r\left(r\frac{(a^2 - 4b)}{4}\right)^2 + ra\left(r\frac{(a^2 - 4b)}{4}\right) - rb.$$

Since $c = a^4 - 4b$ then it results

$$f_{D_2}^2(x_{c_2}, r, a, b) = \frac{-r^3c^2 + 4r^2ac - 16rb}{16}$$

Therefore, if we make $f_{D_2}^2(x_{c_2}, r, a, b) = x_{2,2}^*$ we obtain the following result

$$\frac{-r^3c^2 + 4r^2ac - 16rb}{16} = \frac{ra - 1 - \sqrt{(ra - 1)^2 - 4r^2b}}{2r}$$

By simplifying and factoring the previous inequality, the result is:

$$(cr^{2} - 2ar - 8)(c^{3}r^{3} - 6ac^{2}r^{2} + 4c(8b + a^{2} + 2c)r - 8a(2c + 8b - a^{2})) = 0.$$

We take the quadratic factor $(cr^2 - 2ar - 8) = 0$ and solving for r, we obtain:

$$r = \frac{2a \pm \sqrt{4a^2 + 32c}}{2c} = \frac{a \pm \sqrt{a^2 + 8c}}{c}$$

By substituting $a = \zeta_1 + \zeta_2$, $b = \zeta_1 \zeta_2$, $c = a^4 - 4b$ and $r = \gamma \beta_2$ and by simplifying, we have

$$\gamma \beta_2 = \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{(\zeta_2 - \zeta_1)^2}.$$

Analogously

$$\gamma = \frac{(\zeta_1 + \zeta_2) \pm \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2}$$

Notice that the solution

$$\gamma = \frac{(\zeta_1 + \zeta_2) - \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2} < 0.$$

Therefore, do not satisfies that $\gamma \in (0, 4] \in \mathbb{R}$. Therefore, we take

$$\gamma = \frac{(\zeta_1 + \zeta_2) + \sqrt{9(\zeta_1 + \zeta_2)^2 - 32\zeta_1\zeta_2}}{\beta_2(\zeta_2 - \zeta_1)^2},$$

and by substituting $\beta_2 = 4$, $\zeta_1 = \frac{1}{2}$ and $\zeta_2 = 1$, we obtain $\gamma = \frac{3 + \sqrt{17}}{2}$, which is the value that γ must take for $\mathcal{A}(\gamma)$ to be an invariant set.

Proposition 21. The bimodal map (4.5) exhibits monostability in the subdomain $\zeta_1 \leq x \leq \zeta_2$ when $\gamma \in \left(\frac{3}{2} + \sqrt{2}, \frac{3 + \sqrt{17}}{2}\right]$.

Proof. For this proof, we will proceed to take results from the previous Propositions. From Proposition 18 we know that the bimodal map do not have monostability in $0 \le x \le \frac{\gamma}{4}$ when $\gamma \ge \frac{3+2\sqrt{2}}{2}$. Moreover, we know from the Proposition 19 that there exists a trapping region $\mathcal{A}(\gamma)$ belongs to

 $\zeta_1 \leq x \leq \zeta_2$ when $\frac{3+2\sqrt{2}}{2} < \gamma < \frac{3+\sqrt{17}}{2}$ and finally, from Proposition 20 we know that when $\gamma = \frac{3+\sqrt{17}}{2}$ the bimodal map has an invariant set $\mathcal{A}(\gamma)$. Therefore, since $\gamma \in \left(\frac{3}{2} + \sqrt{2}, \frac{3+\sqrt{17}}{2}\right]$ then, there exists a trapping region or an invariant set $\mathcal{A}(\gamma)$ allowing to the bimodal map has monostability in $\zeta_1 \leq x \leq \zeta_2$.

Since the map f_{D_1} no longer satisfies Proposition 18 but does satisfy Proposition 21, for any initial condition $x_0 \in [0, 1]$, the orbit either converges to or belongs to the trapping region $\mathcal{A}(\gamma)$, as defined by (4.2), within the subdomain $\zeta_1 \leq x \leq \zeta_2$, resulting in monostability. From Figure 4.17, we can see that when $\gamma = \frac{3 + \sqrt{17}}{2}$, there exists monostability in I = [0, 1]satisfying Proposition 20 and, consequently, Proposition 21.



Figure 4.17: Cobweb diagram for the bimodal map f_D with $\gamma = \frac{3 + \sqrt{17}}{2}$ using the initial condition $x_0 = 0.0357 \in [0, 1]$.

Proposition 22 If $\gamma \in (0,4) \in \mathbb{R}$ then the system (4.5) has a trapping region $\mathcal{A} = I = [0,1] \subset \mathbb{R}$.

Proof. To demonstrate this proposition, we need to show that there exists a trapping region if $\gamma \in (0, 4) \in \mathbb{R}$. For reach this goal, we consider the following four intervals $I_0 = [0, x_{c_1}]$, $I_1 = \left[x_{c_1}, \frac{1}{2}\right]$, $I_2 = \left[\frac{1}{2}, x_{c_2}\right]$ and $I_3 = [x_{c_2}, 1]$, such that $\mathcal{A} = \bigcup_{i=0}^3 I_i$, where $x_{c_1} = \frac{1}{4}$ and $x_{c_2} = \frac{3}{4}$ are the critical points of the subdomains $0 \le x < \frac{1}{2}$ and $\frac{1}{2} \le x \le 1$, respectively. For any $x \in \left[0, \frac{1}{2}\right)$ the equation of this modal is

$$f_{D_1}(x,\gamma) = 4\gamma x \left(\frac{1}{2} - x\right).$$

Therefore, if we evaluate each interval I_i for $i = \{0, 1\}$ in f_{D_1} we obtain that $f_{D_1}([0, x_{c_1}], \gamma) = \left[0, \frac{\gamma}{4}\right]$ and $f_{D_1}\left(\left[x_{c_1}, \frac{1}{2}\right], \gamma\right) = \left[0, \frac{\gamma}{4}\right]$. Now, for any $x \in \left[\frac{1}{2}, 1\right]$ the equation for this modal is

$$f_{D_2}(x,\gamma) = 4\gamma \left(x - \frac{1}{2}\right) (1-x)$$

if we evaluate we obtain the following results $f_{D_2}\left(\left[\frac{1}{2}, x_{c_2}, \gamma\right]\right) = \left[0, \frac{\gamma}{4}\right]$ and $f_{D_2}\left(\left[x_{c_2}, 1\right], \gamma\right) = \left[0, \frac{\gamma}{4}\right]$ and since $\gamma \in (0, 4)$ specifically $\gamma < 4$, then,

$$f_{D_1}(I_0,\gamma) = f_{D_1}(I_1,\gamma) = f_{D_2}(I_2,\gamma) = f_{D_2}(I_3,\gamma) = [0,1)$$
 for $\gamma \in (0,4)$.

Analogously

$$f_{D_1}(I_0 \cup I_1, \gamma) = f_{D_2}(I_2 \cup I_3, \gamma) = f_D\left(\bigcup_{i=0}^3 I_i, \gamma\right) = f_D(\mathcal{A}, \gamma) = [0, 1),$$

and since $[0,1) \subset [0,1]$. Therefore, we have that $f_D(\mathcal{A},\gamma) \subset \mathcal{A}$.



Figure 4.18: Cobweb diagram for the bimodal map f_D with $\gamma = 3.6$ using the initial condition $x_0 = 0.8491 \in [0, 1]$.

Figure 4.18 depicts the cobweb diagram for the parameter value $\gamma = 3.6$, according to Proposition 22. The bimodal map (4.5) exhibits monostability within $I = [0, 1] \subset \mathbb{R}$.

Proposition 23 The bimodal map (4.5) has an invariant set $\mathcal{A} = I = [0,1] \subset \mathbb{R}$ when $\gamma = 4$.

Proof. This proof is similar to the previous one, again, we need to show that there exists an invariant set if $\gamma = 4$. For this, we consider the following four intervals $I_0 = [0, x_{c_1}]$, $I_1 = \begin{bmatrix} x_{c_1}, \frac{1}{2} \end{bmatrix}$, $I_2 = \begin{bmatrix} \frac{1}{2}, x_{c_2} \end{bmatrix}$ and $I_3 = [x_{c_2}, 1]$, such that $\mathcal{A} = \bigcup_{i=0}^3 I_i$, where $x_{c_1} = \frac{1}{4}$ and $x_{c_2} = \frac{3}{4}$ are the critical points of the subdomains $0 \le x < \frac{1}{2}$ and $\frac{1}{2} \le x \le 1$, respectively. For any $x \in \left[0, \frac{1}{2}\right)$ the equation of this modal is

$$f_{D_1}(x,\gamma=4) = 16x\left(\frac{1}{2} - x\right)$$

Therefore, if we evaluate each interval I_i for $i = \{0, 1\}$ in f_{D_1} we obtain that $f_{D_1}([0, x_{c_1}], \gamma = 4) = [0, 1]$ and $f_{D_1}\left(\left[x_{c_1}, \frac{1}{2}\right], \gamma = 4\right) = [0, 1]$. Now, for any

 $x \in \left[\frac{1}{2}, 1\right]$ the equation for this modal is

$$f_{D_2}(x, \gamma = 4) = 16\left(x - \frac{1}{2}\right)(1 - x),$$

if we evaluate we obtain the following results $f_{D_2}\left(\left[\frac{1}{2}, x_{c_2}\right], \gamma = 4\right) = [0, 1]$ and $f_{D_2}\left([x_{c_2}, 1], \gamma = 4\right) = [0, 1]$ and since $\gamma \in (0, 4)$. By the previous results, we obtain

$$f_{D_1}(I_0, \gamma = 4) = f_{D_1}(I_1, \gamma = 4) = f_{D_2}(I_2, \gamma = 4) = f_{D_2}(I_3, \gamma = 4) = [0, 1].$$

Analogously

$$f_{D_1}(I_0 \cup I_1, \gamma = 4) = f_{D_2}(I_2 \cup I_3, \gamma = 4)$$
$$= f_D\left(\bigcup_{i=0}^3 I_i, \gamma = 4\right) = f_D(\mathcal{A}, \gamma = 4) = [0, 1].$$

Therefore, we have that $f_D(\mathcal{A}, \gamma = 4) = \mathcal{A}$.



Figure 4.19: Cobweb diagram for the bimodal map f_D with $\gamma = 4$ using the initial condition $x_0 = 0.9340 \in [0, 1]$.

Figure 4.19 depicts the cobweb diagram for the parameter value $\gamma = 4$, according to the Proposition 23. The bimodal map (4.5) exhibits monostability within $I = [0, 1] \subset \mathbb{R}$.

Proposition 24 The bimodal map (4.5) shows monostability in I = [0, 1] if $\gamma \in (0, 4] \subset \mathbb{R}$.

Proof. Notice that when $0 < \gamma < \frac{1}{2}$, then there exists an invariant point $x_{1,2}^*$ inside of I, more precisely, in the subdomain $0 \le x < \frac{1}{2}$. When $\frac{1}{2} < \gamma < \frac{3}{2}$, now the invariant point is $x_{1,1}^*$ and belongs to $0 \le x < \frac{1}{2}$. By the Proposition 18, if $\gamma \in \left[2, \frac{3+2\sqrt{2}}{2}\right]$, there is only one invariant set $\mathcal{A} \in \left[0, \frac{\gamma}{4}\right] \subset I$. From the Propositions 19 and 20, if $\frac{3+2\sqrt{2}}{2} < \gamma \le \frac{3+\sqrt{17}}{2}$ now the only trapping region or invariant set is in $\frac{1}{2} \le x \le 1$. Finally, notice that when $\gamma \in \left(\frac{3+\sqrt{17}}{2}, 4\right]$, for any $x_0 \in I$ the orbit remains in I, given rise to a trapping region or invariant set. Therefore, observe that if $0 < \gamma \le 4$ the bimodal map (4.5) always shows monostability in I.



Figure 4.20: Bifurcation diagram of the bimodal map f_D given by (4.5) with respect to the parameter $\gamma \in (0, 4] \in \mathbb{R}$.

Figure 4.20 shows the corresponding bifurcation diagram for this case study, highlighting a clear relationship between the behaviors found in the cobweb diagram for each case. Below we briefly describe the different behaviors observe in the monoparametric family of multimodals maps given by (4.5).

- For $\gamma \in \left(0, \frac{1}{2}\right)$ the bimodal map (4.5) exhibits an invariant point at $x_{1,2}^* = 0$.
- For $\gamma \in \left[2, \frac{3}{2} + \sqrt{2}\right)$ the bimodal map (4.5) has an invariant set $\mathcal{A} \subset I$, where $\mathcal{A} = \left[0, \frac{\gamma}{4}\right]$ and $\frac{1}{2} \leq \frac{\gamma}{4} < \frac{3+2\sqrt{2}}{8}$. Moreover, the bimodal map (4.5) exhibits monostability in the interval $0 \leq x \leq \frac{\gamma}{4}$ when $\gamma \in \left[2, \frac{3}{2} + \sqrt{2}\right)$.
- When $\gamma \in \left(\frac{3}{2} + \sqrt{2}, \frac{3 + \sqrt{17}}{2}\right)$ the bimodal map (4.5) has a trapping region $\mathcal{A}(\gamma)$ in the interval $\frac{1}{2} \leq x \leq 1$, and it exhibits monostability.
- When $\gamma = \frac{3 + \sqrt{17}}{2}$ the bimodal map (4.5) has an invariant set $\mathcal{A}(\gamma)$ in the interval $\frac{1}{2} \le x \le 1$.
- For $\gamma \in (0, 4)$ the bimodal map (4.5) has a trapping region $\mathcal{A} = [0, 1] \subset \mathbb{R}$.
- For $\gamma = 4$ the bimodal map (4.5) has a invariant set $\mathcal{A} = [0, 1] \subset \mathbb{R}$.
- For $\gamma \in (0, 4] \subset \mathbb{R}$ the bimodal map (4.5) always shows monostability in $I = [0, 1] \in \mathbb{R}$.

Chapter 5

Conclusion

In the present work, the necessary conditions for ensuring the existence of monostability and bistability in a bimodal map were provided through propositions. These propositions were analytically demonstrated using tools commonly employed in the study of discrete dynamical systems. Subsequently, numerical simulations were conducted to observe the existence of monostability and bistability.

Some results presented were introduced for regular and irregular partitions, therefore this fact allows us to present the statements in terms of the set of points $\{\zeta_0, \zeta_1, \zeta_2\}$. We found that the location and stability of the fixed points of the bimodal map (3.1) in the interval $[\zeta_0, \zeta_1)$ are controlled by the parameter β_1 when it takes values in an interval determined by ζ_0 and ζ_1 . In a similar way, it is possible to determine the location and stability of the fixed points of the bimodal map (3.1) in the interval $[\zeta_1, \zeta_2]$ by β_2 when it takes values in an interval determined by ζ_1 and ζ_2 .

A trapping region or an invariant set in the first modal appears if the modal value $f_{D_1}(x_{c_1})$ is less than the value of the fixed point $x_{2,2}^*$ of the second modal. And in the second modal a trapping region or an invariant set appears if the second iteration of the critical point of the second modal $f_{D_2}^2(x_{c_2})$ is equal to or greater than the value of the fixed point $x_{2,2}^*$. Therefore, the fixed point $x_{2,2}^*$ is crucial to exhibits monostability or bistability in the bimodal map f_D given by (3.1).

We introduced three parametric families of the bimodal maps by considering the parameters β_1 , β_2 and γ as $f_D(x, \beta_1)$, $f_D(x, \beta_2)$ and $f_D(x, \gamma)$, respectively. In the first two families $f_D(x, \beta_1)$ and $f_D(x, \beta_2)$, the value of first modal and the value of the second modal change, respectively, while the value of the other modal remains fixed, then by controlling the values of the first and second modal allow these families to exhibit monostability or bistability in the interval I. In the last family $f_D(x, \gamma)$, the parameter γ does not allow different values for the first and second modal and this restricts the system to only exhibits monostability in the interval I. Within the development, the respective Cobweb diagrams were shown for each case, the bifurcation diagrams were also developed, with the objective of observing the behavior of the different families of the difference map when their parameters take different values.

Bibliography

- E. Campos-Cantón, R. Femat, and A. N. Pisarchik. A family of multimodal dynamic maps. *Communications in Nonlinear Science and Numerical Simulation*, 16(9):3457–3462, 2011.
- [2] Fred Attneave. Multistability in perception. *Scientific American*, 225(6):62–71, 1971.
- [3] A. Sahnoune and D. Berkani. On the correlation of chaotic signals generated by multimodal skew tent map. Signal, Image and Video Processing, 12(7):1273–1278, 2018.
- [4] M. García-Martínez, I. Campos-Cantón, E. Campos-Cantón, and S. Celikovský. Difference map and its electronic circuit realization. *Nonlinear Dynamics*, 74(3):819–830, November 2013.
- [5] Chunbiao Li, Julien Clinton Sprott, Wen Hu, and Yujie Xu. Infinite multistability in a self-reproducing chaotic system. *International Jour*nal of Bifurcation and Chaos, 27(10):1750160, September 2017.
- [6] Timothy S. Gardner, Charles R. Cantor, and James J. Collins. Construction of a genetic toggle switch in escherichia coli. *Nature*, 403(6767):339– 342, 2000.
- [7] Debsankha Manik, Marc Timme, and Dirk Witthaut. Cycle flows and multistability in oscillatory networks: an overview. arXiv preprint arXiv:1611.09825, 2016.
- [8] MEE Abashar and SSEH Elnashaie. Multistability, bistability and bubbles phenomena in a periodically forced ethanol fermentor. *Chemical Engineering Science*, 66(23):6146–6158, 2011.

- [9] C. Li, W. Hu, J. C. Sprott, and X. Wang. Multistability in symmetric chaotic systems. *European Physical Journal: Special Topics*, 224(8):1493–1506, July 2015.
- [10] Mingshu Peng, Zhonghao Jiang, Xiaoxia Jiang, Jiping Hu, and Youli Qu. Multistability and complex dynamics in a simple discrete economic model. *Chaos, Solitons & Fractals*, 41(2):671–687, 2009.
- [11] Hui Chang, Yuxia Li, Guanrong Chen, and Fang Yuan. Extreme multistability and complex dynamics of a memristor-based chaotic system. *International Journal of Bifurcation and Chaos*, 30(08):2030019, 2020.
- [12] George CMA Ehrhardt, Matteo Marsili, and Fernando Vega-Redondo. Phenomenological models of socioeconomic network dynamics. *Physical Review E*, 74(3):036106, 2006.
- [13] Alexander N. Pisarchik and Ulrike Feudel. Control of multistability. *Physics Reports*, 540(4):167–218, July 2014.
- [14] Michael K. Smits, Leendert W. Hamoen, and Oscar P. Kuipers. Bistability in the bacillus subtilis k-state (competence) system. *Molecular Microbiology*, 56(3):957–967, 2005.
- [15] Till Kuhlbrodt, Sven Titz, Ulrike Feudel, and Stefan Rahmstorf. A simple model of seasonal open ocean convection: Part ii: Labrador sea stability and stochastic forcing. *Ocean Dynamics*, 52:36–49, 2001.
- [16] Geert Lenderink and Reindert J Haarsma. Variability and multiple equilibria of the thermohaline circulation associated with deep-water formation. Journal of Physical Oceanography, 24(7):1480–1493, 1994.
- [17] Harish Parthasarathy and Suresh Babu. Three-mode controllable master-slave monostable multivibrators using cfoa-based circuits. *IET Circuits, Devices & Systems*, 8(6):432–438, 2014.
- [18] Joseph Mark and Sarah Fisher. Bistability from double phosphorylation in signal transduction. FEBS Journal, 273(17):3572–3580, 2006.
- [19] S. González-Salas, B. B. Cassal-Quiroga, J. Tuxpan, and E. Campos. Dynamics of multimodal families of m-modal maps. *Complexity*, 2022:4340794, 2022.

- [20] Tom M Apostol. Mathematical analysis; 2nd ed. Addison-Wesley series in mathematics. Addison-Wesley, Reading, MA, 1974.
- [21] H.K. Khalil. Nonlinear Systems. Macmillan Publishing Company, 1992.
- [22] S.H. Friedberg, A.J. Insel, and L.E. Spence. *Linear Algebra*. Pearson Education, 2014.
- [23] Richard Holmgren. A First Course in Discrete Dynamical Systems. Springer Science & Business Media, 2000.
- [24] Bernd SW Schröder. Mathematical analysis: a concise introduction. John Wiley & Sons, 2007.
- [25] R. Larson. Cálculo de una variable. Educación (McGraw-Hill). McGraw-Hill Interamericana de España S.L., 2010.
- [26] Lawrence Perko. Differential equations and dynamical systems, volume 7. Springer Science & Business Media, 2013.
- [27] A. Celletti. Stability and Chaos in Celestial Mechanics. Springer Praxis Books. Springer Berlin Heidelberg, 2010.
- [28] Kathleen T Alligood, Tim D Sauer, James A Yorke, and David Chillingworth. Chaos: an introduction to dynamical systems. *SIAM Review*, 40(3):732–732, 1998.
- [29] Robert L. Devaney. A First Course in Chaotic Dynamical Systems: Theory and Experiment. CRC Press, 2018.
- [30] J. Bahi and C. Guyeux. Discrete Dynamical Systems and Chaotic Machines: Theory and Applications. Chapman & Hall/CRC Numerical Analysis and Scientific Computing Series. CRC Press, 2013.
- [31] K.T. Alligood, T.D. Sauer, and J.A. Yorke. *Chaos: An Introduction to Dynamical Systems*. Textbooks in Mathematical Sciences. Springer New York, 2000.
- [32] C. Robinson. Dynamical Systems: Stability, Symbolic Dynamics, and Chaos. Studies in Advanced Mathematics. CRC Press, 1998.